

Citation: Collins CR, Hackett F, Atid J, Tan MSY, Blackman MJ (2017) The *Plasmodium falciparum* pseudoprotease SERA5 regulates the kinetics and efficiency of malaria parasite egress from host erythrocytes. PLoS Pathog 13(7): e1006453. https://doi.org/10.1371/journal.ppat.1006453

Editor: Oliver Billker, Wellcome Trust Sanger Institute, UNITED KINGDOM

Received: March 25, 2017

Accepted: June 7, 2017

Published: July 6, 2017

Copyright: © 2017 Collins et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the paper and its Supporting Information files.

Funding: This work was supported by funding to MJB from the Francis Crick Institute (https://www. crick.ac.uk/) which receives its core funding from Cancer Research UK (FC001043; https://www. cancerresearchuk.org), the UK Medical Research Council (FC001043; https://www.mrc.ac.uk/), and the Wellcome Trust (FC001043; https://wellcome. ac.uk/), as well as Wellcome Trust ISSF2 funding **RESEARCH ARTICLE**

The *Plasmodium falciparum* pseudoprotease SERA5 regulates the kinetics and efficiency of malaria parasite egress from host erythrocytes

Christine R. Collins¹, Fiona Hackett¹, Jonathan Atid¹, Michele Ser Ying Tan¹, Michael J. Blackman^{1,2}*

1 Malaria Biochemistry Laboratory, The Francis Crick Institute, London, United Kingdom, 2 Department of Pathogen Molecular Biology, London School of Hygiene & Tropical Medicine, London, United Kingdom

* mike.blackman@crick.ac.uk

Abstract

Egress of the malaria parasite *Plasmodium falciparum* from its host red blood cell is a rapid, highly regulated event that is essential for maintenance and completion of the parasite life cycle. Egress is protease-dependent and is temporally associated with extensive proteolytic modification of parasite proteins, including a family of papain-like proteins called SERA that are expressed in the parasite parasitophorous vacuole. Previous work has shown that the most abundant SERA, SERA5, plays an important but non-enzymatic role in asexual blood stages. SERA5 is extensively proteolytically processed by a parasite serine protease called SUB1 as well as an unidentified cysteine protease just prior to egress. However, neither the function of SERA5 nor the role of its processing is known. Here we show that conditional disruption of the SERA5 gene, or of both the SERA5 and related SERA4 genes simultaneously, results in a dramatic egress and replication defect characterised by premature host cell rupture and the failure of daughter merozoites to efficiently disseminate, instead being transiently retained within residual bounding membranes. SERA5 is not required for poration (permeabilization) or vesiculation of the host cell membrane at egress, but the premature rupture phenotype requires the activity of a parasite or host cell cysteine protease. Complementation of SERA5 null parasites by ectopic expression of wild-type SERA5 reversed the egress defect, whereas expression of a SERA5 mutant refractory to processing failed to rescue the phenotype. Our findings implicate SERA5 as an important regulator of the kinetics and efficiency of egress and suggest that proteolytic modification is required for SERA5 function. In addition, our study reveals that efficient egress requires tight control of the timing of membrane rupture.

Author summary

Malaria, a disease that kills hundreds of thousands of people each year, is caused by a single-celled parasite that grows in red blood cells of infected individuals. Following each round of parasite multiplication, the infected red cells are actively ruptured in a process called egress, releasing a new generation of parasites. Egress is essential for progression to to the London School of Hygiene & Tropical Medicine. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

clinical disease, but little is known about how it is controlled. In this work we set out to address the function in egress of a *Plasmodium falciparum* protein called SERA5, an abundant component of the vacuole in which the parasite grows. We show that parasites lacking SERA5 (or lacking both SERA5 and a closely-related protein called SERA4) undergo accelerated but defective egress in which the bounding vacuole and red cell membranes do not rupture properly. This impedes the escape and subsequent replication of the newly-developed parasites. We also show that modification of SERA5 by parasites proteases just prior to egress is important for SERA5 function. Our results show that SERA5 is a 'negative regulator' of egress, controlling the speed of the pathway that leads to disruption of the membranes surrounding the intracellular parasite. Our findings increase our understanding of the molecular mechanisms underlying malarial egress and show that efficient egress requires tight control of the timing of membrane rupture.

Introduction

Malaria is caused by protozoan parasites of the genus *Plasmodium*. Sporozoites introduced into the human host by the mosquito vector migrate to the liver where the parasite replicates to produce merozoites. These are released into the bloodstream, initiating the asexual blood stage of the infection in which the parasite goes through multiple rounds of intraerythrocytic replication and host cell destruction, producing gradually increasing parasitaemia that eventually leads to clinical disease. Like many intracellular pathogens, the parasite replicates within an intracellular membrane-bound compartment called a parasitophorous vacuole (PV). Replication is by schizogony, in which formation of a multinucleated schizont occurs before a budding or segmentation process generates the individual daughter merozoites. In *Plasmodium falciparum*, the most widespread agent of fatal malaria, each intraerythrocytic replication cycle takes ~48 h, with production of 16 or more merozoites per schizont.

During schizont development, a number of soluble parasite proteins accumulate within the PV lumen. Amongst the best characterized of these is a set of proteins belonging to the serine rich antigen (SERA) family, so named due to the presence of 27 or more consecutive Ser residues within the first-recognized member of the family, variously called 111 K antigen, P126, P140, SERA or SERP I but now referred to as SERA5 in P. falciparum (PlasmodDB ID PF3D7_0207600). A unifying feature of the SERA proteins, first noticed in SERA5 by Higgins et al. [1] then confirmed by x-ray crystallographic determination of the P. falciparum SERA5 central domain by Hodder and colleagues [2], is their possession of a central domain homologous to papain-like cysteine peptidases (clan CA, family C1). SERA family members are found in all *Plasmodium* genomes examined [3], and whilst the number of genes varies depending on the species, in all cases they fall into two classes: those that encode a Cys residue at the position equivalent to the nucleophilic Cys25 of papain (Cys-type); and those that possess a Ser codon at this position (Ser-type). Gene disruption analysis of the 9 P. falciparum SERA genes suggested that only two, SERA5 (Ser-type) and SERA6 (Cys-type), are important in the haploid asexual blood stage parasites [4-6], implying crucial roles for SERA5 and SERA6 in this clinically relevant part of the parasite life cycle. Very recent work using conditional mutagenesis has confirmed that disruption of the *P. falciparum SERA6* gene is lethal [7].

Release (egress) of daughter merozoites from the infected erythrocyte has long been known to be sensitive to cysteine protease inhibitors, including the selective covalent modifier transepoxysuccinyl-L-leucylamido(4-guanidino)butane (E64) (e.g. [8]). The resemblance of the SERA proteins to cysteine proteases, together with their subcellular localisation in the PV in

both blood stages [4, 9, 10] and liver stages [11], has spurred interest in the possibility of the SERA proteins playing a role in egress. In support of this, disruption of a SERA family member that is highly expressed in mosquito stages of the parasite life cycle produced a defect in release of sporozoites from oocysts, structures on the basal surface of the insect midgut in which sporozoite biogenesis occurs [12]. Additionally, SERA5, which is the most abundantly-expressed family member in P. falciparum blood stages, was shown in early studies to be subjected to extensive proteolytic processing that coincided temporally with and was dependent upon egress [13–16], suggesting a link between SERA5 function and egress. Subsequent work in P. falciparum has uncovered the mechanism underlying this proteolytic processing. Minutes before egress, activation of a parasite cGMP-dependent protein kinase called PKG leads to the discharge of specialised merozoite organelles called exonemes which contain a subtilisin-like serine protease called SUB1 [17-19]. Upon secretion into the PV lumen, SUB1 cleaves both SERA5 and SERA6 at 2 or 3 discrete positions, releasing their central papain-like domains [10, 17]. The available evidence suggests that SERA6 possesses proteolytic activity that is activated by this processing [10]. In contrast, SERA5 has a non-enzymatic role in the parasite, since mutations predicted to abolish catalytic activity (e.g. substitution with Ala of the putative nucleophilic Ser596) had no phenotypic effect, whilst similar approaches failed to obtain viable parasites possessing a disrupted SERA5 gene [6]. Thus there is considerable evidence that SERA5 and SERA6 have important functions in the parasite. However, whether any member of the SERA family has a role in blood stage egress and what that role may be, is unresolved.

Here we have used a conditional genetic approach combined with selective pharmacological tools to provide the first experimental evidence that SERA5 regulates egress in asexual blood stages of *P. falciparum*. Remarkably, rather than acting as a mediator of egress as previously suspected, we show that SERA5 enhances egress efficiency by acting as a negative regulator of the kinetics of egress.

Results

Efficient conditional ablation of SERA5 expression using the DiCre system

In previous work [20] we described the production of a transgenic P. falciparum 3D7-derived clone called 1G5DiCre (here abbreviated to 1G5DC). These parasites possess a modified, partially recodonised (chimeric) SERA5 gene encoding the wild type SERA5 amino acid sequence, followed by a single chromosomally-encoded *loxP* site and an integrated DiCre expression cassette. The latter drives constitutive expression of two individual, enzymatically inactive domains of Cre recombinase, each fused to a different rapamycin (RAP)-binding protein, such that addition of RAP induces heterodimerization of the proteins and Cre recombinase activity [21–23]. SERA5 protein expression by 1G5DC parasites is at wild type levels and the parasites replicate and egress normally in culture. Importantly, production of the 1G5DC clone included a step in which the human dihydrofolate reductase (hdhfr) drug resistance marker used to select for the desired homologous recombination event was removed from the genome by DiCre-mediated excision. The 1G5DC parasites are therefore fully sensitive to the antifolate drug WR99210 [20]. This recycling step allowed us to reuse the *hdhfr* selectable marker in a second gene targeting step (Fig 1A) in which a further loxP site was introduced into the parasite genome by targeted homologous recombination at the 3' end of the upstream SERA4 gene, reconstituting the SERA4 gene whilst effectively floxing the entire chimeric SERA5 locus. Two of the resulting transgenic parasite clones, called floxSERA5-1B6 and floxSERA5-3B6 were selected for further analysis. Examination by diagnostic PCR confirmed the expected genomic architecture (Fig 1A), whilst pulse-treatment of synchronised, newly-invaded (ring-stage)



IFA of integrant clone floxSERA5-1B6

Fig 1. DiCre-mediated conditional disruption of *P. falciparum* **SERA5 expression.** (A) Strategy for conditional deletion of the *SERA5* gene. Transgenic *P. falciparum* clone 1G5DC contains a wild-type *SERA4* gene (white boxes, introns indicated as gaps) upstream of a fully functional chimeric *SERA5* gene (recodonised sequence, hatched). This is followed by a single *loxP* site (black arrow head) and the DiCre expression cassette (grey box). The targeting constructs contained ~1.0 kb (pSERA4loxPa) or ~1.4 kb (pSERA4loxPb) of 3' *SERA4* sequence to drive integration of the entire construct by single-crossover homologous recombination. In both cases the targeting sequence extended to and included the *SERA4* stop codon and was followed by the 3' UTR of the *P. berghei* dihydrofolate reductase (*Pbdf*) gene to ensure correct regulation of the introduced human dihydrofolate reductase-thymidylate synthase (*hdhfr*) selection cassette which confers resistance to the antifolate WR99210. DiCre-mediated recombination was predicted to excise the entire sequence between the *loxP* sites, including the *SERA5* gene. Positions of hybridisation of primers used for diagnostic PCR analysis of integration and excision events are shown as coloured arrows. Primer identities are: a, S4_F4; b, S4_DS_R1; c, CAM5'_R3; d, hsp86_3'_R1 (see <u>S1_Table</u> for sequences of primers used in this study). Insert, diagnostic PCR analysis of genomic DNA from the parental 1G5DC *P. falciparum* clone and integrant clones, confirming the predicted or control parental 1G5DC and

integrant *P. falciparum* clones, confirming the predicted DiCre-mediated excision events. The expected sizes of the PCR amplicons specific for the intact or excised locus are indicated. (C) IFA of mature schizonts of control (DMSO-treated) and RAP-treated integrant clone floxSERA5-1B6 ~44 h following treatment. No SERA5-specific signal was detectable in the majority (~98%) of the RAP-treated population. Scale bar, 5 µm. (D) Western blot of mature schizonts of the integrant clones ~44 h following RAP-treatment or mock treatment, showing near complete loss of the SERA5 signal. As a loading control, blots were probed with antibodies to both SERA5 and the major merozoite surface protein MSP1. Quantitation of the residual SERA5 signal in the RAP-treated parasites (S1 Fig) indicated that this was likely due to lack of DiCre-mediated gene excision in a small proportion (~2%) of the parasite population.

https://doi.org/10.1371/journal.ppat.1006453.g001

parasites with 100 nM RAP for just 1 h resulted in the expected DiCre-mediated excision event, rapidly deleting the entire chimeric *SERA5* coding sequence with high efficiency within a single erythrocytic cycle (Fig 1B).

To assess the effects of SERA5 gene deletion on SERA5 protein expression, mature flox-SERA5-1B6 and floxSERA5-3B6 schizonts were examined by indirect immunofluorescence (IFA) at the end of the first erythrocytic cycle (~44 h) following RAP-treatment. No SERA5-specific signal was detectable by IFA in most of the RAP-treated floxSERA5-1B6 and flox-SERA5-3B6 schizonts (Fig 1C). Exhaustive microscopic examination showed apparently normal SERA5 expression in ~2% of schizonts (S1A Fig), likely representing a small population of parasites in which gene excision had not occurred. This was confirmed by Western blot analysis of schizonts of the RAP-treated floxSERA5-1B6 and floxSERA5-3B6 populations, indicating an overall >95% reduction of SERA5 expression (Fig 1D and S1B Fig). RAP-treatment had no detectable effect on expression of an unrelated merozoite protein, MSP1 (Fig 1C and 1D and S1B Fig), nor on expression of the SERA4 and SERA6 genes, which flank the SERA5 gene (S1C Fig) [3, 5]. Schizont morphology and merozoite numbers in the Δ SERA5 parasites were normal by light microscopy of Giemsa-stained preparations (S1D Fig), and detailed visual examination of schizont SDS extracts fractionated on Coomassie-stained gels showed no detectable effects of RAP-treatment on the total parasite protein profile except for the noticeable absence of a ~120 kDa species identified by Western blot as full-length SERA5 (S1E Fig). These results were consistent with the genetic data, indicating specific, rapid and efficient DiCre-mediated disruption of SERA5 expression.

SERA5 is important but dispensable for blood stage *P. falciparum* growth *in vitro*

To establish the effects of SERA5 loss on parasite growth, we compared the replication rates of mock-treated and RAP-treated floxSERA5-1B6 and floxSERA5-3B6 clones, using the parental 1G5DC clone as a control. No morphological or growth differences were evident in the ~44 h immediately following RAP-treatment (referred to as cycle 0), and all the parasites matured to schizont stage at the same rate (Fig 1C, S1A and S1D Fig) with no differences in the number of merozoites produced per schizont (S1D Fig; also see below). However, a ~50% reduction in the number of intracellular parasites was evident in the RAP-treated floxSERA5-1B6 and flox-SERA5-3B6 cultures by the middle of the next cycle, and further monitoring into cycle 2 revealed a clear replication defect in these cultures (Fig 2A). Over a more prolonged period, periodic examination of the RAP-treated floxSERA5-1B6 and floxSERA5-3B6 cultures by IFA and diagnostic PCR indicated a time-dependent increase in the proportion of SERA5-expressing parasites in the cultures (Fig 2B and 2C), suggesting that the initially small population of non-excised parasites gradually overgrew the cultures, likely as a result of a selective advantage conferred on them by the replication defect displayed by the Δ SERA5 parasites. However, even after 12 erythrocytic growth cycles (24 days), Δ SERA5 parasites were still detectable in the RAP-treated cultures (Fig 2C, lower right-hand panels), proving that whilst loss of SERA5 expression severely impacted the rate of parasite replication, it did not completely abolish it.



Fig 2. Loss of SERA5 expression results in severely reduced replication and invasion rates. (A) Relative replication rates of clones floxSERA5-1B6 and floxSERA5-3B6 compared to the parental 1G5DC clone, over the first 3 erythrocytic cycles following mock

treatment (-) or RAP-treatment (+) at ring stage in cycle 0. Cultures were fed daily by replacing the medium, but were not passaged by addition of fresh erythrocytes. Parasitaemia values were determined by FACS at trophozoite stage as described in Materials and Methods. Data are compiled from 6 independent biological replicate experiments. Error bars, ±SEM. RAP-treatment of the floxSERA5-1B6 and floxSERA5-3B6 clones was calculated to result in a mean reduction in replication rate per cycle of 53±8%. (B) Diagnostic PCR analysis of genomic DNA isolated from parasites harvested at intervals following initiation of one of the assays depicted in (A) except that in this case the cultures were maintained for a more prolonged period (12 erythrocytic cycles), requiring regular passaging by dilution into medium containing fresh erythrocytes as described in Materials and Methods. Prolonged maintenance of the RAP-treated parasites resulted in selective expansion of the small number of non-excised parasites in the starting population, with concomitant substantial loss of parasites harbouring the excised locus (only just detectable by cycle 12). Primers used for amplification of the intact and excised loci were as in Fig 1, whilst primers specific for the SERA4 locus were used for the control PCR. (C) IFA of purified schizonts, confirming outgrowth of SERA5-positive parasites in the RAP-treated floxSERA5-3B6 population, consistent with the PCR results. Whereas at cycle 0 (~44 h following RAP-treatment) the majority of the RAP-treated population were SERA5-negative, by cycle 12 very few ΔSERA5 parasites were left in the RAP-treated population (arrowed). However, the continued presence of the ΔSERA5 parasites showed that they can continue to replicate for prolonged periods in vitro. Scale bar, 20 µm. (D) Loss of SERA5 reduces ring formation but has no effect on intracellular development. Typical invasion assay showing formation and development of new ring stage parasites following rupture of RAP-treated or mock-treated floxSERA5-1B6 and floxSERA5-3B6 schizonts. Schizonts (11% parasitaemia) were incubated with fresh erythrocytes in static or shaking cultures for just 4 h before removal of residual intact schizonts by centrifugation over Percoll cushions. Parasitaemia was then determined by FACS following further incubation of the ring cultures in static conditions (early schizont stage; 30 h). RAP treatment produced a consistent reduction in ring formation, but had no or little impact on subsequent intracellular development. Similar results were obtained in three independent experiments.

https://doi.org/10.1371/journal.ppat.1006453.g002

To define the point(s) in the erythrocytic cycle affected by loss of SERA5 expression, parallel cultures of synchronous RAP-treated or mock-treated floxSERA5-1B6 or floxSERA5-3B6 mature schizonts at identical parasitaemia (11%) at the end of cycle 0 were allowed to undergo egress and invasion for just 4 h, under either normal static conditions or whilst continuously and vigorously shaken. Following removal of residual intact schizonts by centrifugation over Percoll cushions, the newly-formed ring-stage parasites were enumerated at once by FACS. Subsequent intracellular development (in static culture) of these parasites was also monitored by FACS approximately 30 h later, as well as by microscopic examination. As shown in Fig 2D, shaking the cultures during the invasion period increased ring formation in all cases, implying more efficient egress and/or invasion as previously observed by others [24], but production of rings was consistently reduced in the RAP-treated cultures compared to their mock-treated counterparts. However, once formed, these rings developed similarly over the course of the second intraerythrocytic cycle, irrespective of their provenance. Together with the previous evidence that SERA5-null parasites form morphologically normal schizonts, these results convincingly suggested that the reduction in long-term replication rate associated with loss of SERA5 was primarily or exclusively due to a defect not in intracellular parasite growth, but at the transition between schizont maturation and formation of new ring stage parasites.

To quantify the effects of SERA5 loss on long-term parasite growth in the absence of parasites still expressing SERA5, fresh floxSERA5-1B6 and floxSERA5-3B6 cultures were RAP- or mock-treated then identically diluted to obtain densities of <100 parasites/ml and transferred to flat-bottomed microwell plates. Examination of the plaques visible in these wells after 14–16 days (which form as a result of localised erythrocyte destruction in the static erythrocyte layers [7]) confirmed a growth defect in the Δ SERA5 parasites, with mock-treated floxSERA5-1B6 and floxSERA5-3B6 parasites producing significantly larger and more numerous plaques than their RAP-treated counterparts (Fig 3A). Parasites from 2 separate cloning wells of the RAPtreated floxSERA5-1B6 and floxSERA5-3B6 parasites, each of which contained only a single small plaque, were individually expanded. The resulting parasite clones (called 2F8_ Δ SERA5 and C6_ Δ SERA5) were confirmed as Δ SERA5 by Western blot (Fig 3B). They were then compared to the parental clones (not RAP-treated) in growth assays (Fig 3C). The results showed that loss of SERA5 results in a ~28% decrease in replication rate at each erythrocytic cycle. This difference in replication rates was rather less than the ~50% observed in static cultures immediately following *SERA5* gene excision (Fig 2A and 2D), suggesting the possibility of some adaptation of the 2F8_ Δ SERA5 and C6_ Δ SERA5 clones during the ~12 week period required for their cloning by limiting dilution and subsequent expansion. Nonetheless, it could be concluded unambiguously from these results that SERA5 is important for efficient blood stage parasite growth but is not essential for long-term viability *in vitro*.



Fig 3. ΔSERA5 parasite clones show a replication defect but SERA5 is not essential for asexual blood stage growth in vitro. (A) Scatter plots showing plaque numbers and distribution of plaque sizes obtained following treatment of the parental 1G5DC clone or conditional knockout clones floxSERA5-1B6 and floxSERA5-3B6 with DMSO (control, mock-treated) or RAP to induce disruption of the SERA5 gene. In each case plaque numbers were counted from a total of 24 microwells. Note that plaque numbers were lower in the RAP-treated floxSERA5-1B6 and floxSERA5-3B6 clones (n = 60 and 20 respectively) than in the mock-treated samples (n = 94 and 86 respectively). Plaque dimensions in scanned images of the plates were quantified using the Lasso tool of Photoshop CS6 (Adobe). Horizontal bars, mean plaque area ±1 SD. Brackets and asterisks indicate parasite populations for which statistically different plaque sizes were obtained (Student's two-tailed t-test). The difference in plaque size for the +/- RAP-treated parental 1G5DC parasites were not significant (p-value = 0.1564). (B) Western blot confirmation of Δ SERA5 parasite clones obtained from plaque assay wells containing a single plaque. Clone 2F8_ Δ SERA5 (derived from RAP-treated clone floxSERA5-1B6) and C6_ΔSERA5 (derived from RAP-treated clone floxSERA5-3B6) were expanded, then schizont extracts probed with antibodies to SERA3, SERA4, SERA5 and SERA6. (C) Growth assay showing relative replication rates of Δ SERA5 *P. falciparum* clones 2F8_ Δ SERA5 and C6_ Δ SERA5 compared to their parental floxSERA5-1B6 and floxSERA5-3B6 clones (not RAP-treated), over 3 erythrocytic cycles. Cultures were fed daily by replacing the medium, but were not passaged by addition of fresh erythrocytes. Parasitaemia values were determined by FACS as described in Materials and Methods. Data are averaged from 3 independent biological replicate experiments. Error bars, \pm SEM. Relative to the parental parasites, the Δ SERA5 clones showed a mean reduction in replication rate per cycle of 28±3%.

https://doi.org/10.1371/journal.ppat.1006453.g003

SERA5 null parasites undergo accelerated rupture but inefficient egress

To gain insights into the defect underlying the reduced rates of new ring formation displayed by ΔSERA5 parasites, highly synchronous preparations of the floxSERA5-1B6 and floxSERA5-3B6 clones were RAP- or mock-treated at ring stage, then examined as they reached the point of egress. To enhance the synchrony of egress in these experiments, the purified mature schizonts (~43 h post-treatment) were transferred for 4–6 h into medium containing one or other of the PKG inhibitors 4-[2-(4-fluorophenyl)-5-(1-methylpiperidine-4-yl)-1H-pyrrol-3-yl] pyridine (compound 1), or (4-[7-[(dimethylamino)methyl]-2-(4-fluorphenyl)imidazo[1,2- α]pyridine-3-yl]pyrimidin-2-amine (compound 2) both of which potently but reversibly block parasite development at a stage just preceding egress [18, 25]. Short-term treatment of schizonts with either compound results in the accumulation of 'stalled' highly mature schizonts in which the PVM is intact though often porous [26]. Subsequent wash-out of the inhibitors releases the egress block, allowing the schizonts to proceed to rupture within minutes and facilitating time-lapse video microscopic examination of multiple egress events over a relatively short period [18, 26, 27]. As shown in Fig 4A-4C, as well as S1 Movie and S2 Fig, the RAPtreated floxSERA5-1B6 and floxSERA5-3B6 parasites displayed a dramatic egress phenotype characterized by abnormally rapid membrane rupture and inefficient dispersal of the released merozoites, which often initially formed asymmetric 'clusters' apparently due to them being transiently trapped within partially ruptured bounding membranes. The Δ SERA5 merozoites generally gradually dispersed from these clusters over a period of several minutes, proving that both the PV and host erythrocyte membranes had undergone rupture (Fig 4C and S2 Fig). Indeed, in many cases, PVM rupture clearly occurred shortly before erythrocyte membrane rupture in both mock-treated and Δ SERA5 parasites. This was indicated by transient rounding-up of the schizonts with swelling of the PV, then sudden loss of differential interference contrast of the cell and increased visibility and motility of the internal merozoites (S1 Movie), as previously observed in both P. falciparum [26, 28-30] and the zoonotic malarial species P. knowlesi [31]. The total proportion of schizonts that underwent rupture over the imaging periods did not differ between parallel mock and RAP-treated cultures, confirming that egress was not inhibited by loss of SERA5 expression. In both mock and RAP-treated schizonts, egress was often accompanied by sudden lateral displacement of the schizont by as much as $5-10 \mu m$, perhaps indicating rapid expulsion of the cytosolic contents of the cell upon the release of the osmotic pressure proposed by some to be responsible for the pre-egress PV swelling [28, 32]. The time interval between washing away the PKG inhibitors and initiation of egress in control (mock-treated) parasites differed somewhat between the PKG inhibitors, with a rather longer delay in the case of compound 1 (minimum delay to egress ~13 mins) compared to compound 2 (minimum time to egress \sim 7.5 min). However, in all cases the Δ SERA5 parasites exhibited significantly accelerated membrane rupture compared to similarly treated controls. This was confirmed by Coomassie-staining and Western blot analysis, which showed more rapid appearance of soluble parasite-derived proteins in the supernatants of RAP-treated flox-SERA5-1B6 schizonts released from a compound 2-mediated egress block than in the mocktreated samples (Fig 4D). In contrast, Western blot analysis of the schizonts themselves showed no discernible difference in the rates of proteolytic processing of two established SUB1 substrates, SERA6 and MSP1 (S3 Fig). Importantly, the abortive egress phenotype and formation of merozoite 'clusters' was also observed in mature Δ SERA5 schizonts that had not been synchronised by treatment with compounds 1 or 2, confirming that the phenotype was not an artefact associated with prior exposure to the PKG inhibitors (S2 Movie). Time-lapse microscopic examination of the 2F8_ Δ SERA5 and C6_ Δ SERA5 clones compared with RAP-treated parental 1G5DC parasites (S3-S5 Movies) confirmed the premature egress phenotype in the



Fig 4. Loss of SERA5 results in accelerated schizont rupture but defective egress. (A) Stills from timelapse DIC microscopic imaging showing stages in rupture of mock (DMSO)-treated (control) and RAP-treated

(ASERA5) schizonts of P. falciparum clone floxSERA5-1B6. Movies were started exactly 4.5 min following removal of the reversible PKG inhibitor compound 2 (time following washing away the inhibitor is indicated). ΔSERA5 parasites underwent accelerated membrane rupture which had often already started even at the commencement of imaging (white arrows) with only gradual dispersal of the merozoites, unlike the 'explosive' egress and merozoite scattering typical of control parasites. Egress of the control parasites did not commence in this experiment until ~9 min (an example is indicated, black arrow). Scale bar, 20 µm. See also S1 Movie and S2 Fig. (B) Quantitative profiling of the timing of membrane rupture in the control and Δ SERA5 parasites. Whereas egress in the control parasites only rarely took place before 9 min, much of the membrane rupture evident in the ΔSERA5 parasites had already occurred by that point. Data were collated from visual examination of frames from 3 separate videos each of mock and RAP-treated clone floxSERA5-1B6 (total number of egress events: mock-treated, 178, RAP-treated, 179;). Time of egress is recorded to the nearest 0.5 min. Note that because imaging began 4.5 min following washing, all rupture events in the RAP-treated population that had already taken place at the beginning of the video microscopy are recorded as occurring at 4.5 min. Time to egress statistics were calculated for the RAP-treated parasites (mean 7.9 min, SD 3.2 min) and for the control parasites (mean 13.7 min, SD 3.3. min), with a two-tailed unpaired t-test revealing the difference to be extremely significant (t = 16.8268, d.f. = 355, p < 0.0001). (C) Stills from time-lapse DIC microscopic images of individual control and ΔSERA5 schizonts, showing the very different characteristics of merozoite dispersal in the period immediately following the point of schizont rupture (the first still was imaged 10 seconds prior to rupture). Scale bar, 10 µm. (D) Coomassie-stained, SDS PAGE-fractionated and Western blot analysis of supernatants from mock and RAP-treated (ΔSERA5) floxSERA5-1B6 schizonts following release from a compound 2 block, showing more rapid release of schizont soluble proteins, including proteolytically processed forms of SUB1 [59] from the ΔSERA parasites. See also S3 Fig.

https://doi.org/10.1371/journal.ppat.1006453.g004

 Δ SERA5 clones and showed that these parasites had not undergone major compensatory phenotypic alterations during the extended culture period (~12 weeks) required for their isolation by limiting dilution cloning and expansion. This phenotype was still evident even after extended continuous culture (>16 months) of these parasite clones. It was concluded that loss of SERA5 results in accelerated schizont rupture but defective egress, and that this likely explains the replication defect evident in the Δ SERA5 parasites.

SERA5 null parasites are not intrinsically defective in host erythrocyte membrane poration or vesiculation

Elegant previous studies by Glushakova et al. and others in P. falciparum have demonstrated that, in the seconds just preceding host red blood cell membrane breakage at egress, the membrane becomes permeabilized or 'porated', allowing leakage of cellular contents and the ingress of extracellular molecules [26, 32, 33]. The latter includes the F-actin-binding peptide phalloidin which, if introduced into the surrounding culture medium, is able just before egress to access and label the actin protofilaments of the cytoskeleton that underlies the host erythrocyte cell membrane. By allowing schizont rupture in the presence of fluorescent phalloidin, poration can be detected by microscopy as the sudden appearance of a fluorescent signal around the host cell circumference [32, 33]. In addition, phalloidin binding to the ruptured host erythrocyte cytoskeleton immediately after egress enables visualisation of the state of the fragmented membranes, which generally undergo extensive vesiculation [28, 32]. Host erythrocyte membrane rupture (but not PVM rupture) is selectively blocked by the cysteine protease inhibitor E64 but poration of the membrane is not [26, 32], so the combined use of E64 and fluorescent phalloidin provides a simple means of reliably observing the onset and efficiency of poration, since the porated schizonts do not rupture. To examine whether loss of SERA5 expression impacts on poration and fragmentation of the erythrocyte membrane, compound 2-synchronised, mock or RAP-treated floxSERA5-3B6 schizonts were washed and transferred to fresh medium containing Alexa Fluor 488-labeled phalloidin, with or without E64. Simultaneous time-lapse DIC and fluorescence imaging of the E64-containing cultures showed the gradual appearance of phalloidin-labelled schizonts as the host cell membrane of these cells became porated (Fig 5A). Intriguingly, the appearance of phalloidin labelling in individual



Fig 5. SERA5 does not play a role in host cell membrane poration or vesiculation at egress. (A) Stills from simultaneous time-lapse DIC and fluorescence microscopic imaging showing ingress of Alex Fluor 488-labelled phalloidin into intact mock-treated (control) and RAP-treated (Δ SERA5) schizonts of *P. falciparum* clone floxSERA5-3B6 in the

presence of E64. Movies were started exactly 4.5 min following removal of compound 2 (time following washing away the inhibitor is indicated). Scale bar, 20 μ m. (B) Quantitative profiling of the timing of phalloidin labelling in the control and Δ SERA5 schizonts. Data were collated from visual examination of frames from 3 separate videos each of mock and RAP-treated parasites (total number of phalloidin labelling events: RAP-treated, 56; mock-treated, 51). Time to labelling is indicated to the nearest 0.5 min. Statistics were calculated for the Δ SERA5 parasites (mean 23.7 min, SD 7.9 min) and the control parasites (mean 27.2 min, SD 7.6. min), with a two-tailed unpaired *t*-test revealing the difference to be just significant (t = 2.3260, d.f. = 105, p = 0.0219). (C) Stills from simultaneous time-lapse DIC and fluorescence microscopic imaging of control and Δ SERA5 floxSERA5-386 schizonts in the presence of fluorescent phalloidin but the absence of E64, showing vesiculation and phalloidin labelling of the disrupted host cell membranes following egress. Note that the released merozoites are slightly out of focus in these images in order to focus on the vesiculated membranes. Scale bar, 10 μ m. See also S4 Fig.

https://doi.org/10.1371/journal.ppat.1006453.g005

schizonts was always temporally coincident with PVM rupture, the latter being clearly evident by DIC imaging as the sudden loss of differential interference contrast of the schizont and increased visibility and mobility of the intracellular merozoites (S6 Movie). Of additional interest, compared to the kinetics of egress in parallel samples lacking E64 which progressed rapidly to rupture with kinetics as described in Fig 4, phalloidin labelling in the presence of E64 was substantially delayed in both control and Δ SERA5 samples (compare Fig 5B with Fig 4B). This is the first evidence to our knowledge that although E64 does not prevent PVM rupture, it delays the process. Whilst the rate of appearance of phalloidin-labelled schizonts in the presence of E64 was slightly faster in the ΔSERA5 schizont population (Fig 5A and 5B), this was only just significant, in contrast to the clear premature rupture phenotype displayed by the Δ SERA5 parasites in the absence of E64. The total proportion of schizonts labelled with phalloidin within a ~35 min period following removal of the compound 2 block in the presence of E64 was similar for the Δ SERA5 and control cultures (Fig 5B). Time-lapse DIC examination of E64-containing cultures in the absence of phalloidin detected no significant difference between the timing of PV rupture in the control and ΔSERA5 populations (S4 Fig). In cultures containing phalloidin but lacking E64, extensive vesiculation of the residual host cell membranes was observed in all cases following egress (Fig 5C). Collectively, these results showed that SERA5 does not play a direct role in mediating host cell membrane poration or vesiculation. However, the fact that in the presence of E64 the rate of progress to PVM rupture and host cell membrane poration in the Δ SERA5 parasites was very similar to that of control parasites, implied a potential link between SERA5 function and the protein target(s) of E64. This point is further discussed below.

Simultaneous disruption of both the *SERA4* and *SERA5* genes produces an egress phenotype indistinguishable from that resulting from *SERA5* disruption

The genes encoding the Ser-type SERA family members SERA1-5 in P. falciparum lie in a head-to tail tandem array on chromosome 2 [3, 5]. This, together with the high degree of synteny between SERA loci in all Plasmodium species examined, suggests that the Ser-type SERA genes arose through gene duplication events, the number of which varied between different *Plasmodium* species [3]. Phylogenetic analyses have suggested that all the Ser-type SERA proteins may mediate similar functions [3, 5]. However, expression levels vary widely, with both transcriptional and proteomic data indicating that in *P. falciparum* SERA5 is the most highly expressed, whilst SERA4 is the second most abundantly expressed [4, 34, 35] and is dispensable in blood stages [5]. In the light of our success in flanking the SERA5 gene on the 1G5DC background with *loxP* sites using a single-crossover homologous recombination approach, we decided to employ a modified strategy to simultaneously flox both the SERA5 and SERA4 genes with the aim of examining the consequences of disrupting both genes simultaneously.



Fig 6. Simultaneous disruption of both the SERA4 and SERA5 genes results in a replication defect similar to that produced by SERA5 disruption alone. (A) Western blot analysis of the C10_ Δ SERA4/5 and G8_ Δ SERA4/5 *P. falciparum* clones (as well as the 1G5DC parental clone as a control), confirming loss of expression of both SERA4 and SERA5 in the double knockout parasites (note that the 1G5DC loading control blots are the same as shown in Fig 3, since all the Western blots were performed at the same time). (B) Growth assay showing relative replication rates of the Δ SERA4/5 *P. falciparum* clones compared to the single knockout 2F8_ Δ SERA5 and C6_ Δ SERA5 clones, over 3 erythrocytic cycles. Cultures were fed daily by replacing the medium, but were not passaged by addition of fresh erythrocytes. Parasitaemia values were determined by FACS. Data are averaged from 5 independent biological replicate experiments. Error bars, ±SEM. See also S5 Fig.

https://doi.org/10.1371/journal.ppat.1006453.g006

As shown in S5 Fig, a construct (pSERA3loxP) designed to insert a *loxP* site immediately downstream of the *SERA3* gene successfully integrated in the expected manner into the 1G5DC genome. Treatment with RAP of a parasite clone (called floxSERA4/5-B52) harbouring the integrated construct resulted in excision of both the *SERA4* and *SERA5* genes as shown by diagnostic PCR analysis (S5 Fig). The resulting parasites were cloned by limiting dilution and two clones obtained. Examination by Western blot of mature schizonts of these clones (called C10_ Δ SERA4/5 and G8_ Δ SERA4/5) confirmed the absence of both SERA4 and SERA5 but unaltered expression of SERA3 and SERA6, as expected (Fig 6A).

To assess the effects of simultaneous loss of both SERA4 and SERA5 on parasite egress, we examined the timing and morphology of egress of the floxSERA4/5-B52 parasites following mock-treatment or treatment with RAP. As shown in S7 Movie and S8 Movie, the Δ SERA4/5 parasites displayed an accelerated but defective egress phenotype identical to that of Δ SERA5 parasites. Growth assays comparing the replication rates of the Δ SERA4/5 clones with the Δ SERA5 clones showed no significant differences in replication rates over the course of 3 erythrocytic cycles (Fig 6B). It was concluded that simultaneous loss of both SERA4 and SERA5 produces a defect in egress no more severe than that resulting from loss of SERA5 alone.

Episomal expression of the wild-type *SERA5* gene, but not a mutant refractory to SUB1-mediated cleavage, rescues the growth defect associated with *SERA5* knockout

The very similar phenotype displayed by the Δ SERA5 and the Δ SERA4/5 parasites, together with the conditional nature of the gene disruption strategy and the apparently unaltered expression of the flanking SERA3 and SERA6 genes in both sets of knockouts, gave us a high degree of confidence that the observed phenotype was a direct result of manipulation of the SERA4/5 loci and not due to unintended genetic alterations in the modified parasites. To confirm this and to investigate the structural requirements for SERA5 function, we sought to rescue the Δ SERA5 phenotype by genetic complementation. For this, we introduced into $2F8 \Delta SERA5$ parasites a transgene construct (called pDC2 mC sgS5) designed for episomal expression of the SERA5 gene under the control of endogenous flanking 5' sequence likely to include the genomic SERA5 promoter (Fig 7A). To reliably identify the transgene product, the SERA5 transgene was modified by the introduction of an internal mini TAP tag sequence incorporating a hemagglutinin (HA3) epitope tag, that we had previously shown does not interfere with SERA5 function in the parasite, despite abolishing the SUB1 site 2 processing site [6]. The pDC2_mC_sgS5 construct also contained a cassette for constitutive expression of cytoplasmic mCherry to facilitate identification of parasites harbouring the construct. In parallel, 2F8_ASERA5 parasites were independently transfected with a mutant of the same episome, called pDC2_mC_sgS5mut, in which the SERA5 sequence encoding the site 1 SUB1 processing site [17], as well as a site that is cleaved by an anonymous cysteine protease referred to as protease X [6], were modified in a manner designed to block correct cleavage of the transgene product (Fig 7A). A third set of $2F8_{\Delta}SERA5$ parasites were transfected with a similar construct that contained the mCherry expression cassette but lacked the SERA5 expression cassette. Transfected parasites were subjected to selection in the presence of blasticidin. IFA and Western blot analysis of the resulting three blasticidin-resistant lines (called 2F8 ΔSERA5:SER-A5wt, 2F8_ΔSERA5:SERA5mut and 2F8_ΔSERA5:mCherry) showed mCherry expression in all cases (although not in every parasite; see below) and confirmed expression of the transgenic SERA5 proteins in the 2F8 Δ SERA5:SERA5wt and 2F8 Δ SERA5:SERA5mut lines (Fig 7A and 7B), as well as the expected SERA5 processing defect in the mutant line (Fig 7B). To examine the effects of transgene expression on the egress phenotype, egress of synchronous mature schizonts of the 2F8 Δ SERA5:SERA5wt and 2F8 Δ SERA5:SERA5mut parasite lines following release of a compound 2-mediated block was visualised by simultaneous live DIC and fluorescence time-lapse imaging as previously. Substantial variation in the levels of mCherry expression was evident in the imaged schizonts, likely due to heterogeneity in episome segregation as previously noted for episomal plasmids in P. falciparum [36, 37]. However comparison of the fluorescent with the non-fluorescent schizonts in each population clearly showed rescue of the delayed (Fig 7C) and abortive (S9 Movie and S10 Movie) egress phenotype in the 2F8_ΔSERA5:SERA5wt parasites but not in the mutant line. It was concluded that correct proteolytic processing of SERA5 is likely essential for its function in regulating the kinetics of egress.

Discussion

Like all obligate intracellular pathogens, the malaria parasite has evolved efficient molecular strategies to exit from its host cell. In this study we have made inroads into our understanding of the regulatory mechanisms underlying blood stage egress in *P. falciparum*, showing that conditional disruption of the *SERA5* gene, or both the *SERA4* and *SERA5* genes simultaneously, leads to a dramatic egress defect characterised by premature but inefficient membrane





2F8_ΔSERA5. All constructs were designed for constitutive expression of cytoplasmic mCherry, driven by the P. falciparum CAM promoter. Constructs pDC2 mC sqS5 and pDC2 mC sqS5mut additionally incorporated cassettes for expression of a synthetic recodonised SERA5 gene (under control of the SERA5 promoter) modified by inclusion of an internal mini TAP tag which includes the HA3 epitope [6]. Positions of mutations (see S6 Fig for details of mutations) introduced into pDC2_mC_sgS5mut to prevent processing by SUB1 and protease X [6] are indicated (arrowheads). bsd, blasticidin deaminase drug resistance cassette. Right-hand side: IFA of the blasticidin-resistant parasite lines selected following transfection of the SERA5-null parasite clone 2F8_DSERA5 with the constructs. Reactivity with the anti-HA3 mAb 3F10 was observed only in mature schizonts of the 2F8_ΔSERA5:SERA5wt and 2F8_ΔSERA5:SERA5mut lines, in a pattern typical of PV localisation. For clarity, the merged images do not include the DAPI signal. Scale bar, 10 µm. (B) Western blot analysis of purified mature schizonts of the transfected parasite lines, either following incubation for 4 h in the presence of the PKG inhibitor compound 2 (C2 blocked) or 40 min after washing away of compound 2 to allow egress of the majority of the schizonts (egressed). This involves discharge of SUB1 and proteolytic processing of SERA5 and the merozoite surface protein MSP1 [18, 27]. HA3-tagged SERA5 was expressed only in the 2F8_ΔSERA5:SERA5wt and 2F8_DSERA5:SERA5mut lines. As expected, proteolytic processing of the protein was defective in the 2F8_DSERA5: SERA5mut line, leading to a species (asterisked) that was much larger than the HA3-tagged processing product of wildtype SERA5 (indicated and labelled). In contrast correct SUB1-mediated processing of MSP1 to form the MSP1₈₃ processing product [27] occurred similarly in all three parasite lines. (C) Quantitative profiling plot of the timing of egress in the 2F8_ASERA5:SERA5wt and 2F8_ASERA5:SERA5mut parasite lines. In each case, time to egress of mCherrypositive schizonts (red) following washing away of a compound 2 block is shown compared with that of mCherry-negative parasites (grey) in the same fields. The mCherry-positive schizonts were defined as those with a mean fluorescence intensity value of ≥30 under standardised fluorescence imaging conditions (see Materials and Methods). Time-lapse imaging began precisely 4.5 min following washing, and data were collated from visual examination of frames from 3 separate videos each of the two lines. Time to egress statistics were calculated for the mCherry-positive 2F8_ΔSERA5: SERA5wt parasites (mean 17.08 min, SD 4.52 min) and for the mCherry-negative 2F8 ASERA5:SERA5wt parasites (mean 12.0 min, SD 6.42 min), with a two-tailed unpaired t-test revealing the difference to be extremely significant (t = 3.5985, d.f. = 59, p < 0.001). In contrast, similar analysis of the time to egress statistics for the mCherry-positive 2F8 ΔSERA5:SERA5mut parasites (mean 13.28 min, SD 5.91 min) and mCherry-negative 2F8 ΔSERA5:SERA5mut parasites (13.69 min, SD 5.57 min) revealed the difference to be non-significant (t = 0.2397, d.f. = 45, p = 0.812). See also S9 Movie and S10 Movie.

https://doi.org/10.1371/journal.ppat.1006453.g007

rupture. The released daughter merozoites disseminate slowly, apparently due to their being retained transiently within residual, incompletely fragmented bounding membranes. The result is a reduction in invasion of new host erythrocytes. The indistinguishable phenotypes of the Δ SERA5 and Δ SERA4/5 mutants is consistent with the notion of functional redundancy amongst the Ser-type *SERA* genes and previous evidence that loss of SERA4 alone is tolerated by the parasite [5].

Our study provides two important insights into the molecular mechanisms underlying blood-stage egress in *P. falciparum*. The first concerns the kinetics of the egress pathway downstream of PKG activation. Our work confirms our previous observations [18, 26] that there is a delay of several minutes between washing away the PKG inhibitors from compound 1 or compound 2-stalled schizonts, and initiation of egress. Here we quantified this period under the conditions used to a minimum of ~13 mins in the case of compound 1 and ~7.5 min for compound 2. We had previously assumed that this delay to egress is primarily a reflection of the rate of dissociation of the respective inhibitor-PKG complexes, and that egress initiates as soon as release of the bound drug allows reactivation of sufficient PKG activity to trigger egress. However our new data force a re-evaluation of that simplistic assumption, because schizont membrane rupture consistently occurs more rapidly in the Δ SERA5 and Δ SERA4/5 mutants. The premature rupture is clearly PKG-dependent, since it is reversibly blocked by PKG inhibitors. It also involves SUB1 discharge (Fig 4C), so our observations reveal for the first time that the delay to egress observed in wild-type parasites following removal of the PKG inhibitors includes a 'lag phase' that is additional to the time required for the parasites to become membrane rupture-competent following PKG activation and SUB1 discharge into the PV (Fig.8). What is the function of this lag phase and how is it controlled?

The answers to these questions may lie in the second important insight provided by our study. The simplest interpretation of the premature rupture phenotype is that SERA5 (and also



Fig 8. Model for role of SERA5 in regulating a lag phase that enables efficient bounding membrane disruption at egress. Schematic model of the role of SERA5 based on the results of this and previous studies. The erythrocyte and PV

membranes of mature schizonts are indicated in red and green respectively. SUB1 in exonemes of mature segmented schizonts is depicted as blue dots, whilst full-length SERA5 is shown as black strands within the PV. Evidence that the PVM becomes porous prior to PKG activation, that PVM disruption results in vesiculated 'rolls' of membrane and that erythrocyte membrane rounding up and/or poration is associated with rapid collapse of the host cell cytoskeleton is from earlier work [26, 32]. Timelines shown to egress are approximate minima and are based on reversible inhibition of egress by compound 2. The final step of host cell membrane rupture is sensitive to E64 in both wild type and Δ SERA5 parasites. However, in the presence of E64 the time to PVM rupture is similar for wild type and Δ SERA5 mutants (not indicated schematically). RBC, red blood cell.

https://doi.org/10.1371/journal.ppat.1006453.g008

perhaps SERA4) is a negative regulator of the egress pathway, acting to maintain the lag phase and thus delay rupture of the PVM and host erythrocyte membranes following SUB1 discharge. The fact that the premature egress phenotype is also associated with a defect in merozoite dissemination suggests that the function of the lag phase is to ensure proper destabilisation of the bounding membranes and/or collapse of the host cell cytoskeleton [26] such that, once they rupture, the merozoites can easily and rapidly escape from the residual membranes. Elegant high speed video microscopy and modelling experiments by Abkarian and colleagues [30] has shown that the scattering of merozoites normally observed at schizont rupture is at least in part due to the elastic properties of the erythrocyte membrane which, upon primary rupture at a single site, undergoes extremely rapid curling and buckling, turning inside-out to literally fling the remaining enclosed merozoites outwards. This demonstrably does not happen in the Δ SERA5 and Δ SERA4/5 mutants. We suggest that in the absence of the SERA5-regulated lag phase, partial membrane rupture occurs before critical intracellular membrane modification events-perhaps those that lead to cytoskeleton collapse [26] or that provide the spontaneous membrane curvature to the erythrocyte membrane thought to be required for the elastic phenomenon described above [30, 38, 39]—have had time to go to completion. Crucially, as soon as the membranes break in the mutants, further membrane modification probably ceases due to the premature leakage and dilution of the effector molecules that mediate these modifications. These would normally function in the confined space of the PV lumen and/or host cell cytosol in the critical few minutes between SUB1 discharge and final egress. These effector molecules may include proteases such as SUB1 itself (which is released into the surrounding medium upon premature rupture of the Δ SERA5 parasites; Fig 4C), as well as SERA6 [10] and possibly host cell-derived calpain-I [40], plus putative pore-forming proteins such as PLP1 [33], which has been proposed as a candidate for mediating the cysteine protease-independent erythrocyte membrane poration that immediately precedes normal egress [32, 33], although recent gene disruption data refute this [41]. The ruptured but incompletely fragmented membranes transiently entrap the merozoites, severely impeding their efficient dissemination (Fig 8). Given the very short invasive half-life of free *P. falciparum* merozoites, which lies somewhere between 90–300 seconds at $37^{\circ}C$ [29, 42–45], it is likely that any delay in the dispersal of free merozoites from the bounding membrane(s) of a ruptured schizont impacts on invasion efficiency. Whilst vigorous shaking during egress increased the efficiency of invasion and ring formation, presumably due to more efficient dispersal of the released merozoites, even this could not completely rescue the phenotype. We propose that the egress defect is the primary cause of the poor replication rates observed in the Δ SERA5 and Δ SERA4/ 5 mutants. However, we cannot rule out other possibilities, including that the Δ SERA5 merozoites may have an intrinsic invasion defect due for example to the absence of surface-bound N- and C-terminal SERA5 processing products, which have been shown to selectively bind merozoites following SUB1-mediated cleavage [9]. Unfortunately, distinguishing between these alternative models is experimentally challenging, and attempts by us to assess the invasive capacity of isolated merozoites from the ruptured Δ SERA5 schizonts were inconclusive.

We can only speculate on the molecular mechanisms underlying SERA5 function. As one of the most abundant PV proteins, SERA5 may simply act as an abundant SUB1 substrate or 'sink', competing with SUB1-mediated cleavage of other key substrates in the moments following SUB1 discharge and thus slowing down the kinetics of the egress pathway. Arguing strongly against this is our inability to discern any changes in the rate of processing of the known SUB1 substrates MSP1 and SERA6 in the Δ SERA5 mutants (S3 Fig). An alternative model rests on the fact that SERA5 bears structural resemblance to a cysteine protease [2] and indeed displays peptidase activity when the 'active site' Ser596 is replaced by a Cys residue [6]. Catalytically inactive enzyme orthologues, including pseudokinases and pseodoproteases, are common, often thought to have evolved from their catalytically active cognate enzymes [46-49] and sometimes existing in regulatory complexes with them [50, 51]. SERA5 may likewise act in concert with a related partner enzyme. A good candidate partner is the egress-related enzyme SERA6, a proposition supported by phylogenetic analyses suggesting that the ancestral gene from which all the Ser-type SERA genes evolved was a Cys-type gene likely corresponding to either SERA6 or SERA7 in *P. falciparum* [3]. However, exhaustive attempts by us to identify interactions between SERA5 and SERA6 or any other candidate partners by co-immunoprecipitation have been unsuccessful; for example, we can readily obtain chromatographically pure SERA5 from parasite extracts in both full-length [17] and processed form [6] under nondenaturing conditions at neutral pH, suggesting that SERA5 does not form high affinity associations with other soluble parasite or host proteins. A third model for SERA5 function, related to the second, is that SERA5 might act by binding and protecting substrates of one or more egress-related proteases, transiently inhibiting cleavage of those substrates. The binding of SERA5 might be released by SUB1 and/or protease X-mediated cleavage of SERA5, explaining the importance of processing for SERA5 function. Notably, either of these pseudoprotease roles for SERA5 would be consistent with our observation in this study that in the presence of E64 the rate of progression to PVM rupture was independent of SERA5 expression. This is because if the function of SERA5 is to control the enzymatic rate of a related cysteine protease against critical physiological substrates, then pharmacological inhibition of that cysteine protease would be expected to negate the effects of loss of SERA5. Elucidating the molecular details of SERA5-mediated regulation will be a key priority of further work.

Prior to this work, all attempts to directly disrupt the *P. falciparum SERA5* gene using conventional targeted homologous recombination had failed [4–6], leading to the conclusion that the gene is indispensable for *in vitro* growth. Our results here, using a robust conditional gene modification system, clearly demonstrate that this is not the case, although we also show that due to their replication defect Δ SERA5 parasites can be rapidly out-competed *in vitro* by wild type parasites. We suspect that the previous lack of success in obtaining SERA5-null parasites was due to the inefficient nature of homologous recombination in *P. falciparum*, combined with the technical difficulties of isolating blood stage mutants with a growth defect because of their selective disadvantage in the presence of wild-type parasites. We expect that future work using DiCre and similar efficient conditional approaches will enable the isolation of many malarial mutants lacking expression of genes previously considered to be essential. As in this case, the resulting phenotypes will provide valuable insights into the biology of this important pathogen.

Materials and methods

Reagents and antibodies

The antifolate drug WR99210 was from Jacobus Pharmaceuticals (New Jersey, USA). Blasticidin, E64 and rapamycin were from Sigma. Rapamycin was used as described previously [20]. The PKG inhibitor compound 1 was a kind gift of David Baker, London School of Hygiene & Tropical Medicine, whilst compound 2 was kindly provided by Dr Simon Osborne, MRC Technology, London NW7 1AA, UK. Stocks of both drugs were stored in dry DMSO at -20°C, and were used throughout at final concentrations of 2 μ M (compound 1) and 1 μ M (compound 2). Monoclonal antibody (mAb) 89.1, which recognises the merozoite surface protein MSP1, has been described previously [52], as has the human anti-MSP1 mAb X509 [53], rabbit polyclonal antisera to *P. falciparum* SERA5 [6] and SERA6 [10], and a rabbit polyclonal antiserum against *P. falciparum* SUB1 [54]. The anti-SERA5 mAb NIMP.M13 was produced using a previously described method [18], using B-cells from BALB/c mice (bred in the pathogen-free animal facility of the Medical Research Council National Institute for Medical Research, Mill Hill, London) immunised with recombinant full-length SERA5 [6]. Rabbit antibodies specific to relatively non-conserved regions of *P. falciparum* SERA3 and SERA4 (anti-SE3N and anti-SE4N) were kind gifts of Professor Toshihiro Horii, Osaka University, Japan.

P. falciparum culture, transfection and limiting dilution cloning

Asexual blood stages of the DiCre-expressing P. falciparum clone 1G5DC [20] were cultured in an atmosphere of 90% nitrogen, 5% carbon dioxide and 5% oxygen at 37°C in RPMI 1640 medium containing Albumax (Invitrogen) supplemented with 2 mM L-glutamine, and synchronised using standard procedures [55, 56]. Parasite developmental stage and viability was routinely assessed by microscopic examination of Giemsa-stained thin blood films. For transfection of integration constructs, Percoll-enriched synchronised mature schizonts were electroporated with plasmid DNA (10 µg per transfection) using an Amaxa P3 primary cell 4D Nucleofector X Kit L (Lonza) as described [27]. Growth medium was replaced ~20 h post transfection with fresh medium containing 2.5 nM WR99210. Once drug-resistant parasites appeared and displayed robust growth (2-3 weeks post-transfection), they were subjected to repeated cycles of culture for 3 weeks without drug followed by culturing with drug (drug cycling) to select for parasites in which integration into the genome had taken place [56]. Transgenic parasite clones floxSERA5-1B6, floxSERA5-3B6 and floxSERA4/5-B52 were obtained by limiting dilution cloning in round-bottomed wells at a calculated 0.1-0.3 parasite per well as described [56]. Following RAP-treatment of these parasites to induce gene disruption, clones of Δ SERA5 or Δ SERA4/5 parasites were obtained by limiting dilution in flat-bottomed 96-well microplate wells as described [7], plating a calculated 10 parasites per well. Only wells containing single plaques were subsequently used. Once established, all transgenic clones except the parental 1G5DC clone were maintained in medium containing 2.5 nM WR99210.

Immunofluorescence and western blot

For IFA, air-dried thin films of parasite cultures were fixed in paraformaldehyde, permeabilized, then probed with relevant primary antibodies as described previously [10]. Secondary Alexa Fluor 488 or 594-conjugated antibodies specific for human, rabbit or mouse IgG (Invitrogen) were used at a dilution of 1:10,000. Samples were stained with 4,6-diamidino-2-phenylindol (DAPI) for nuclear staining then mounted in Citifluor (Citifluor Ltd., UK). Images were acquired using a Zeiss Axioplan 2 Imaging system (Carl Zeiss, Germany) and AxioVision 3.1 software, using identical exposure conditions for all samples being compared. Western blots were prepared and probed as described previously [57].

Production of transgenic *P. falciparum* for conditional disruption of the *SERA5* or *SERA4* plus *SERA5* genes

A second *loxP* site was introduced into the 1G5DC genome upstream of the *SERA5* locus by targeting it to a site just downstream of the *SERA4* gene. To do this, plasmid pHH1_S4int_US-

loxP was first generated by excising the fragment containing the *loxP* site immediately upstream of the SERA5 targeting region in plasmid pHH1_PreDiCre_A [20] using BamHI and HpaI. This fragment was ligated into pHH1- Δ SERA4 [4] pre-digested with the same restriction enzymes. To produce the targeting sequences, PCR products were amplified from 3D7 genomic DNA using Phusion HF DNA polymerase (NEB) with forward primers S4_F3 or S4_HpaI_ F and reverse primer S4_XhoI_R, generating target regions 1 (~1.4 kb) and 2 (~1 kb), respectively. The PCR products were blunt-ended using T4 DNA polymerase (NEB) then digested with XhoI and cloned into plasmid pHH1_S4int_US-loxP pre-digested with HpaI and XhoI to remove the internal SERA4 sequence, giving rise to plasmids pSERA4loxPa and pSERA4loxPb, respectively. The plasmids were independently transfected into mature 1G5DC schizonts and WR99210-resistant parasites subjected to drug cycling to select for integration into the genome. The resulting parasite lines were screened by diagnostic PCR using primers S4_F4 plus S4_DS_ R1 to detect the unmodified 1G5DC SERA5 locus (giving rise to a 1985 bp fragment) and primers CAM5' R3 plus S4 DS R1 to detect the integrated locus (giving rise to a 1941 bp fragment for target fragment 1, or a 1535 bp fragment for target fragment 2). Limiting dilution cloning of the integrated parasite lines resulted in clones floxSERA5-1B6 (from integration using targeting fragment 2) and floxSERA5-3B6 (from integration using targeting fragment 1).

A transgenic *P. falciparum* line for simultaneous conditional disruption of the *SERA4* and *SERA5* genes was generated by introducing a *loxP* site downstream of the *SERA3* locus in the 1G5DC parasite clone, thus flanking both *SERA4* and *SERA5* genes simultaneously. To do this the *SERA3* targeting region was amplified using forward primer S3_F1 and reverse primer S3_R1 to generate a 1.4 kb product. This was digested with SnaBI and XhoI and cloned into pHH1_S4int_US-loxP, pre-digested with HpaI and XhoI, generating construct pSERA3loxP. This was transfected into 1G5DC schizonts and drug-selected parasites screened for integration using primers CAM5'_R3 and S3_DS_R1 (which produce an amplicon of ~1.7 kb). The presence of the endogenous locus was detected using primers S3_F6 and S3_DS_R1 giving rise to a product of 2.0 kb. Limiting dilution cloning of the integrant parasite population generated parasite clone floxSERA4/5-B52.

For both the *SERA5* and *SERA4/5* gene disruption experiments, following treatment of parasites with RAP (100 nM for 1–4 h at 37°C) the non-excised locus was detected by diagnostic PCR using GoTaq Green (Promega), using primers sgS5_seq4F and hsp86_3'_R1 (amplicon size ~1200 bp) whilst the excised locus was detected using primers CAM5'_R3 and hsp86_3'_ R1 (amplicon size 580 bp). The ΔSERA5 *P. falciparum* clones 2F8_ΔSERA5 and C6_ΔSERA5 were selected from RAP-treated conditional knockout floxSERA5-1B6 and floxSERA5-3B6 respectively. The ΔSERA4/5 clones C10_ΔSERA4/5 and G8_ΔSERA4/5 were similarly obtained by limiting dilution cloning of RAP-treated clone floxSERA4/5-B52.

Parasite growth and plaque assays

Parasitaemia measurements by FACS were as described previously [6]. Briefly, parasites recovered at various time-points were stained with the fluorescent vital stain hydroethidine. As negative controls, uninfected erythrocytes in culture medium were stained and processed in the same way. Parasitaemia was calculated using the FACSCalibur flow cytometer (Becton Dickson) as described previously [58]. Briefly, cultures to be analysed were initially screened using forward and side scatter parameters and gated for erythrocytes. From this gated population, the proportion of HE-stained cells in 100,000 cells was determined using the FL2 detector (585/42 nm).

Plaque assays were performed by dispensing parasite cultures in flat-bottomed microplates at a haematocrit of 0.75%, as described [7]. Plates were imaged using a high resolution flat-bed

scanner 14–16 days after setting up the assays. Plaques were counted by visual examination of the images and plaque size quantified using the Lasso tool in Adobe Photoshop CS6. Statistical analysis (linear regression analysis by analysis of covariance and *t*-test) was performed using GraphPad Prism 7 software (CA, USA) or tools available on the GraphPad calculation website (http://www.graphpad.com/quickcalcs/ttest1/). When required, parasites from wells containing a single plaque were expanded by transferring initially to round-bottomed microplate wells, before further expansion into culture flasks.

Complementation plasmid construction and transfection

For transgenic expression of SERA5 or mutants thereof in SERA5-null parasites, a SERA5 expression cassette under the regulation of its native promoter was initially assembled in pHH1_preDiCre_A [20]. This plasmid already contained the 3' fragment of a previously described synthetic recodonised SERA5 gene, called SERA5_{synth} [20]. The 5' region of the SERA5 synthetic gene was excised from N-term-pMK by digesting with Pac1 and blunt ending with T4 DNA polymerase, followed by digestion with SalI. The resulting fragment was cloned into pHH1_preDiCre_A pre-digested with HpaI and SalI giving rise to plasmid pHH1_sgsera5. The SERA5 promoter region was then amplified from 3D7 genomic DNA with primers S5_US_F5 and S5_US_R3 using Phusion HF DNA polymerase and cloned into pHH1_sgsera5 using AfIII and SnaBI sites, giving rise to plasmid pHH1_S5-5'_sgS5. To distinguish the synthetic gene product from that of the endogenous gene, a mini TAP tag was incorporated into the SERA5 coding region just downstream of the P50C (protease X) cleavage site [6] by cloning a SbfI and XmaI fragment including the mini TAP tag from pHH1SERA5chim∆P6-TAP [6] into pHH1_S5-5'_sgS5, generating plasmid pHH1_S5-5'_sgS5_mT. To obtain constitutive expression of mCherry from the SERA5 expression plasmid, the SERA5 expression cassette was then excised from pHH1_S5-5'_sgS5_mT using NotI-HF and SnaBI, blunt-ended with T4 DNA polymerase, and cloned into pDC2-mCherry (a kind gift of Catherine Suarez, The Francis Crick Institute) pre-digested with BamHI and blunt-ended using T4 DNA polymerase, giving rise to the wild-type complementation plasmid pDC2_mC_sgS5.

Mutagenesis of the SUB1 site 1 cleavage of SERA5 was introduced by carrying out two separate PCR reactions using primers sgS5_5'_F_AfIII plus S5_st1II_R, and S5_st1II_F plus PbDT3'_R1. This was followed by an overlapping PCR reaction in which the two PCR products were mixed and amplification carried out using the external primers sgS5_5'_F_AfIII and PbDT3'_R1. The resulting mutated DNA fragment was cloned into pHH1_S5-5'_sgS5_mT using BgIII and AfIII sites, giving rise to plasmid pHH1_S5-5'_sgS5/st1_mT.

The SUB1 site 2 cleavage site of SERA5 was removed when the mini TAP tag was introduced into the *SERA5* sequence [6]. To additionally mutate the protease X (P50C) cleavage site, two separate PCR reactions were carried out using primers S5synF1 plus p50_R_New, and p50_F_New plus PbDT3'_R1. This was followed by an overlapping PCR using primers S5synF1 and PbDT3'_R1. The resulting mutant amplicon was sub-cloned into pHH1_S5-5'_sgS5_mT using BgIII and AvrII giving rise to plasmid pHH1_S5-5'_sgS5/p50+st2_mT. To produce a modified SERA5 gene containing mutations at both the SUB1 sites 1 and 2 as well as the P50C cleavage site, the SUB1 site 1 cleavage site mutation from pHH1_S5-5'_sgS5/ st1_mT was cloned into plasmid pHH1_S5-5'_sgS5/p50+st2_mT using XcmI. The orientation of the insert was confirmed by sequencing, then the mutant DNA fragment sub-cloned into plasmid pDC2_mC_sgS5 using SnaBI and NotI giving rise to the final mutant complementation plasmid pDC2_mC_sgS5mut.

For introduction of the complementation and control constructs into parasites, mature schizonts of the ΔSERA5 *P. falciparum* clone 2F8_ΔSERA5 were electroporated as described

above with 10 μ g of each construct pDC2_mC_sgS5, pDC2_mC_sgS5mut or pDC2_mCherry and maintained initially in medium containing no drug. Approximately 24 h following electroporation, the parasites were transferred into medium supplemented with blasticidin (2 μ g ml⁻¹) then maintained until vigorous growth ensued. All lines were synchronised before use for Western blot, video microscopy or growth assays.

Time-lapse DIC and fluorescence video microscopy

Viewing chambers (internal volume $\sim 80 \ \mu$) for observation of live schizonts were constructed as described [18] by adhering 22 x 64 mm borosilicate glass coverslips to microscope slides with strips of double-sided tape, leaving ~4 mm gaps at each end. Schizont samples were washed then suspended in warm, gassed complete medium, either alone or supplemented where required with E64 (50 µM final concentration), Alexa Fluor 488 phalloidin (Invitrogen; diluted 1:50 from a 200 unit ml⁻¹ stock in methanol), and/or Alexa Fluor 647-conjugated wheat germ agglutinin (Thermo Fisher; diluted 1:1000 from a 1 mg ml⁻¹ stock in phosphatebuffered saline). The schizont suspension was introduced into the pre-warmed chamber, the ends were sealed and the slide transferred to a temperature-controlled microscope stage held at 37°C. Images were taken either on a Zeiss Axio Imager M1 microscope equipped with an EC Plan-Neofluar 100x/1.3 oil immersion DIC objective and an AxioCam MRm camera, or on a Nikon Eclipse Ni-E widefield microscope fitted with a Hamamatsu C11440 digital camera and Nikon N Plan Apo λ 100x/1.45NA oil immersion objective. Images were taken at 5 s intervals over a total of 20-30 min, then annotated and exported as TIFFs, AVI or QuickTime movies using Axiovision 3.1 or Nikon NIS-Elements software. Mean fluorescence intensity values of individual mCherry-expressing schizonts were determined from exported raw image files (TIFF format) as described previously [18], using the Lasso tool and Histogram options of Adobe Photoshop CS6.

Supporting information

S1 Fig. Efficient and selective DiCre-mediated ablation of SERA5 expression in the flox-SERA5-1B6 and floxSERA5-3B6 P. falciparum clones. (A) IFA of mature schizonts of the two integrant parasite clones ~44 h following RAP-treatment. Samples were probed with the anti-SERA5 mAb NIMP.M13 and the anti-MSP1 human mAb X509. Microscopic counts showed that only 1.67±0.13% of the RAP-treated floxSERA5-1B6 parasites and 2.1±0.61% of the RAP-treated floxSERA5-3B6 parasites showed detectable SERA5 expression ($n \ge 1500$ in each case). The fields of view shown here were deliberately selected in order to show examples of the rare residual SERA5-expressing parasites alongside Δ SERA5 parasites. Scale bar, 5 μ m. (B) Western blot quantitation of SERA5 expression levels in mock-treated and RAP-treated floxSERA5-1B6 populations shows an overall reduction in SERA5 expression of >95%. The same mAbs were used to probe as in (A), with the antibody to MSP1 acting as a loading control. Equal volumes (10 µl) of serially-diluted SDS extract from identical numbers of parasites were loaded per track. (C) Western blot showing that DiCre-mediated ablation of SERA5 expression had no discernible effects on expression of SERA4 and SERA6. SERA-specific rabbit antibodies were used to probe the blots. (D) Giemsa-stained images of RAP-treated (Δ SERA5) and mock-treated floxSERA5-1B6 schizonts ~44 h following treatment. Scale bar, 5 µm. (E) Coomassie-stained SDS PAGE gel of SDS extracts of mock-treated or RAP-treated floxSERA5-1B6 schizonts. The only detectable difference was the absence of a ~120 kDa species from the RAP-treated extract, identified by Western blot as SERA5. Positions of prestained molecular mass marker proteins (left-hand track) are indicated. (JPG)

S2 Fig. Loss of SERA5 expression results in accelerated but defective egress. (A) Stills from time-lapse DIC microscopic imaging showing stages in rupture of mock (DMSO)-treated (control) and RAP-treated (ΔSERA5) schizonts of *P. falciparum* clone floxSERA5-1B6. Movies were started exactly 3 min following removal of the reversible PKG inhibitor compound 1 (time following washing away the inhibitor is indicated). The Δ SERA5 parasites underwent accelerated membrane rupture (examples are labelled with white arrows) with only gradual dispersal of the merozoites, unlike the 'explosive' egress typical of control parasites. The total proportion of observed schizonts that underwent rupture in the two populations over the imaging period was 35% (28 of 80) for the mock-treated parasites and 34% (51 of 148) for the RAP-treated population. (B) Quantitation of the timing of membrane rupture in the control and Δ SERA5 parasites. Whereas egress in the control parasites did not take place before 13 min, most of the membrane rupture evident in the Δ SERA5 parasites had already occurred by that point. Time to egress is indicated to the nearest 0.5 min. Data were collated from visual examination of frames from 2-3 videos each of mock and RAP-treated clone floxSERA5-1B6 (total number of egress events: RAP-treated, 51; mock-treated, 28). Time to egress statistics were calculated for the RAP-treated parasites (mean 10.5 min, SD 5.6 min) and for the control parasites (mean 18.8 min, SD 3.2 min), with a two-tailed unpaired t-test revealing the difference to be extremely significant (t = 7.2297, d.f. = 77, p < 0.0001). (C) The residual host red blood cell membrane remains associated with Δ SERA5 merozoite clusters. Two sets of example stills from simultaneous time-lapse DIC and fluorescence microscopic imaging showing ΔSERA5 schizonts surface-labelled with Alexa Fluor 647-conjugated wheat germ agglutinin (WGA; cyan) prior to and following rupture. The presence of fragmented residual host erythrocyte membranes closely associated with the resulting merozoite clusters is clearly visible. Scale bar, 10 µm.

(TIF)

S3 Fig. Loss of SERA5 expression has no effect on the rate of proteolytic processing of SUB1 substrates MSP1 and SERA6. Western blot analysis of control and RAP-treated (Δ SERA5) schizonts of *P. falciparum* clone floxSERA5-1B6, sampled within minutes (indicated) of release from a compound 2-mediated egress block. Blots were probed with antibodies to (A) MSP1 or (B) SERA6, both established SUB1 substrates. SUB1-mediated processing of SERA6 results in its apparent disappearance because the antiserum used does not recognise the processed products [10]. In contrast, the anti-MSP1 antibody used here (mAb 89.1) recognizes both the full-length MSP1 and a doublet corresponding to the processed product MSP1₈₃ (indicated). Loss of SERA5 expression had no detectable effect on the rate of proteolytic processing of either protein. (JPG)

S4 Fig. Loss of SERA5 expression has no effect on the timing of PVM rupture in the presence of the cysteine protease inhibitor E64. (A) Stills from time-lapse DIC microscopic imaging showing time points leading up to and following PVM rupture in a RAP-treated (Δ SERA5) schizont of *P. falciparum* clone floxSERA5-3B6 in the presence of E64 (50 µM). The point of PVM rupture (arbitrarily set to zero seconds) is clearly distinguishable by the sudden loss of differential interference contrast and increased merozoite visibility. Scale bar, 10 µm. (B) Quantitation of the timing of PVM rupture in the control and Δ SERA5 schizonts in the presence of E64 following removal of the reversible PKG inhibitor compound 2. Data were collated from visual examination of frames from time-lapse DIC videos of mock and RAP-treated clone floxSERA5-3B6. Times are indicated to the nearest 0.5 min and all movies were started exactly 4.5 min following washing away the inhibitor. Time to PVM rupture statistics were calculated for the Δ SERA5 parasites (mean 17.0 min, SD 7.3 min) and for the control parasites (mean 16.9 min, SD 6.8 min), with a two-tailed unpaired *t*-test revealing the difference to be not significant (t = 0.0584, d.f. = 121, p = 0.9536). (JPG)

S5 Fig. DiCre-mediated conditional disruption of both the P. falciparum SERA4 and SERA5 genes. (A) Strategy for simultaneous conditional deletion of both the SERA4 and SERA5 genes. Targeting construct pSERA3loxP contains ~1.4 kb of 3' SERA3 sequence to drive integration of the entire construct into the *P. falciparum* clone 1G5DC SERA3 locus by single-crossover homologous recombination. For clarity, the intron-exon structure of the SERA genes is not indicated. The targeting sequence extended to and included the SERA3 stop codon and was followed by the 3' UTR of the P. berghei dihydrofolate reductase (Pbdt) gene to ensure correct regulation of the modified SERA3 gene. Correct integration was expected to reconstitute the gene whilst introducing a *loxP* site downstream of the introduced *hdhfr* selection cassette. DiCre-mediated recombination was predicted to excise the entire sequence between the loxP sites, including both the SERA4 and SERA5 genes. Positions of hybridisation of primers used for diagnostic PCR analysis of integration and excision events are shown as black arrows. Primer identities are: a, S3_F6; b, S3_DS_R1; c, CAM5'_R3; d, hsp86_3'_R1; e, sgS5_seq4F; f, S3_F7; g, S3_R1 (see S1 Table for sequences of all primers used in this study). (B) Diagnostic PCR analysis of genomic DNA from the parental 1G5DC P. falciparum clone and integrant clone floxSERA4/5-B52, confirming the predicted integration event. Expected sizes of the PCR amplicons are indicated. (C) PCR analysis of genomic DNA from RAPtreated or control parental integrant and two Δ SERA4/5 *P. falciparum* clones, confirming the predicted DiCre-mediated excision events. The expected sizes of the PCR amplicons specific for the intact, excised and control locus are indicated. (JPG)

S6 Fig. Mutations introduced into an episomally-expressed SERA5 gene to prevent its proteolytic processing. Schematic representation of the 3D7 *P. falciparum* SERA5 primary structure with the major processing fragments derived from SUB1 cleavage (P47, P56 and P18) indicated. The P56 fragment is trimmed near its C-terminus (to form the P50 terminal processing product; not depicted on this schematic) by further cleavage at the P50C cleavage site, mediated by an unknown cysteine protease called protease X [6]. The wild-type sequence flanking these sites is shown, alongside the modifications made (shown in red) by introduction of a mini TAP tag (major elements of which are also underlined and indicated) as well as introduction of several substitutions at the P50C and SUB1 site 1 cleavage sites. Note that insertion of the mini TAP tag sequence modifies the SUB1 site 2 cleavage site but has previously been shown to be tolerated when introduced into the endogenous *SERA5* gene [6]. All the modifications shown in red were present in the mutant *SERA5* gene expressed in plasmid construct pDC2_mC_sgS5mut, whereas only the mini TAP tag sequence was present in the pDC2_mC_sgS5 construct.

(JPG)

S1 Movie. Δ SERA5 parasites display an egress defect. Mock-treated or RAP-treated flox-SERA5-1B6 parasites were allowed to mature to schizont stage then further incubated for 4 h in the presence of compound 2 (1 μ M). The parasites were washed, resuspended in fresh warm medium without compound 2 and observed by time-lapse DIC microscopy, starting imaging precisely 4.5 min following the first wash. Time from start of imaging is indicated (top lefthand side) in minutes. In the video shown, as well as other videos of the same parasite populations, the total proportion of observed schizonts that ruptured over the imaging period was 67.9% (178 of 262) for the control parasites and 79.2% (179 of 226) for the Δ SERA5 population.

(WMV)

S2 Movie. ΔSERA5 parasites display an egress defect. Synchronous floxSERA5-1B6 parasites were RAP-treated at ring stage, allowed to mature to schizont stage then returned to culture (without PKG inhibitors) to allow further maturation and egress. Samples were then transferred directly to a viewing chamber and observed by time-lapse DIC microscopy. Time from start of imaging is indicated (top left-hand side) in minutes. Clusters of merozoites typical of abortive rupture are obvious, as well as a single abortive egress event (centre of video at ~17.5 min).



S3 Movie. RAP-treatment of the parental 1G5DC *P. falciparum* clone has no effect on egress. A synchronous culture of 1G5DC parasites were RAP-treated at ring stage, allowed to mature to schizont stage then further incubated for 4 h in the presence of compound 1 (2 μ M). The parasites were washed, resuspended in fresh warm medium without compound 1 and observed by time-lapse DIC microscopy, starting imaging precisely 4.5 min following the first wash. Time from start of imaging is indicated (top left-hand side) in minutes. (AVI)

S4 Movie. The Δ SERA5 *P. falciparum* clone 2F8_ Δ SERA5 displays an accelerated and abortive egress defect. 2F8_ Δ SERA5 parasites were allowed to mature to schizont stage then further incubated for 4 h in the presence of the PKG inhibitor compound 2 (1 μ M). The parasites were washed, resuspended in fresh warm medium without compound 2 and observed by time-lapse DIC microscopy, starting imaging precisely 4.5 min following the first wash. Time from start of imaging is indicated (top left-hand side) in minutes. (AVI)

S5 Movie. The Δ SERA5 *P. falciparum* clone C6_ Δ SERA5 displays an accelerated and abortive egress defect. C6_ Δ SERA5 parasites were allowed to mature to schizont stage then further incubated for 4 h in the presence of the PKG inhibitor compound 2 (1 μ M). The parasites were washed, resuspended in fresh warm medium without compound 2 and observed by time-lapse DIC microscopy, starting imaging precisely 4.5 min following the first wash. Time from start of imaging is indicated (top left-hand side) in minutes. (AVI)

S6 Movie. Δ SERA5 parasites are not defective in erythrocyte membrane poration. Synchronous ring stages of the floxSERA5-3B6 clone were RAP-treated, allowed to mature to schizont stage then further incubated for 4 h in the presence of compound 2 (1 μ M). The parasites were washed, transferred to fresh medium containing Alexa Fluor 488-labeled phalloidin plus E64 (50 μ M), then imaged by simultaneous time-lapse DIC and fluorescence imaging. The appearance of phalloidin labelling in individual schizonts (at ~16.0 min in this movie) was always temporally coincident with PVM rupture, visualised as the sudden loss of differential interference contrast of the schizont and increased visibility and mobility of the intracellular merozoites.

(WMV)

S7 Movie. Normal egress by control schizonts of the floxSERA4/5-B52 clone. A synchronous culture of floxSERA4/5-B52 parasites was mock-treated (with 0.1% v/v DMSO) at ring stage and allowed to mature to schizont stage. Schizonts were then further incubated for 4 h in the presence of compound 1 (2 μ M). The parasites were washed, transferred to fresh medium,

and observed by time-lapse DIC microscopy, starting imaging precisely 4.5 min following the first wash. Time from start of imaging is indicated (top left-hand side) in minutes. Normal explosive egress is observed in the majority of schizonts. (WMV)

S8 Movie. Δ **SERA4/5 parasites display an egress defect.** A synchronous culture of flox-SERA4/5-B52 parasites was RAP-treated at ring stage, allowed to mature to schizont stage then further incubated for 4 h in the presence of compound 1 (2 µM). The parasites were washed, transferred to fresh medium, and observed by time-lapse DIC microscopy, starting imaging precisely 4.5 min following the first wash. Time from start of imaging is indicated (top lefthand side) in minutes. Accelerated, abortive rupture is observed starting from ~2 min following beginning of imaging.

(MOV)

S9 Movie. Episomal expression of an epitope-tagged wild-type SERA5 gene rescues the Δ SERA5 abortive egress phenotype. Simultaneous live DIC and fluorescence time-lapse imaging of mature schizonts of the 2F8_ Δ SERA5:SERA5wt parasite line following release of a compound 2-mediated block. The mCherry-positive schizonts (indicating the presence of the complementing episome) display normal egress, whilst abortive egress typical of Δ SERA5 parasites is seen in the other schizonts. (MP4)

S10 Movie. Episomal expression of an epitope-tagged mutant (cleavage-defective) SERA5 gene fails to rescue the Δ SERA5 abortive egress phenotype. Simultaneous live DIC and fluorescence time-lapse imaging of mature schizonts of the 2F8_ Δ SERA5:SERA5mut parasite line following release of a compound 2-mediated block. The mCherry-positive schizonts (indicating the presence of the complementing episome) display abortive egress typical of Δ SERA5 parasites.

(MP4)

S1 Table. Oligonucleotide primer sequences used in this study. (PDF)

Acknowledgments

The authors are indebted to Professor Toshihiro Horii, Osaka University, Japan, for the kind gifts of anti-SE3N and anti-SE4N antibodies specific for *P. falciparum* SERA3 and SERA4. They also thank Catherine Suarez and James Thomas for the gift of plasmid pDC2-mCherry.

Author Contributions

Conceptualization: Christine R. Collins, Michael J. Blackman.

Data curation: Christine R. Collins, Fiona Hackett, Michele Ser Ying Tan, Michael J. Blackman.

Formal analysis: Christine R. Collins, Jonathan Atid, Michael J. Blackman.

Funding acquisition: Michael J. Blackman.

Investigation: Christine R. Collins, Fiona Hackett, Jonathan Atid, Michele Ser Ying Tan, Michael J. Blackman.

Methodology: Christine R. Collins, Fiona Hackett, Michael J. Blackman.

Project administration: Michael J. Blackman.

Resources: Christine R. Collins, Fiona Hackett, Michael J. Blackman.

Supervision: Michael J. Blackman.

Validation: Christine R. Collins, Fiona Hackett, Michael J. Blackman.

Visualization: Christine R. Collins, Fiona Hackett, Jonathan Atid, Michele Ser Ying Tan, Michael J. Blackman.

Writing – original draft: Christine R. Collins, Michael J. Blackman.

Writing - review & editing: Michael J. Blackman.

References

- Higgins DG, McConnell DJ, Sharp PM. Malarial proteinase? Nature. 1989; 340(6235):604. https://doi. org/10.1038/340604a0 PMID: 2671749.
- Hodder AN, Malby RL, Clarke OB, Fairlie WD, Colman PM, Crabb BS, et al. Structural insights into the protease-like antigen Plasmodium falciparum SERA5 and its noncanonical active-site serine. J Mol Biol. 2009; 392(1):154–65. https://doi.org/10.1016/j.jmb.2009.07.007 PMID: 19591843.
- Arisue N, Kawai S, Hirai M, Palacpac NM, Jia M, Kaneko A, et al. Clues to evolution of the SERA multigene family in 18 Plasmodium species. PloS one. 2011; 6(3):e17775. https://doi.org/10.1371/journal. pone.0017775 PMID: 21423628; PubMed Central PMCID: PMCPMC3058004.
- Miller SK, Good RT, Drew DR, Delorenzi M, Sanders PR, Hodder AN, et al. A subset of Plasmodium falciparum SERA genes are expressed and appear to play an important role in the erythrocytic cycle. J Biol Chem. 2002; 277(49):47524–32. https://doi.org/10.1074/jbc.M206974200 PMID: 12228245.
- McCoubrie JE, Miller SK, Sargeant T, Good RT, Hodder AN, Speed TP, et al. Evidence for a common role for the serine-type Plasmodium falciparum serine repeat antigen proteases: implications for vaccine and drug design. Infect Immun. 2007; 75(12):5565–74. https://doi.org/10.1128/IAI.00405-07 PMID: 17893128; PubMed Central PMCID: PMCPMC2168336.
- Stallmach R, Kavishwar M, Withers-Martinez C, Hackett F, Collins CR, Howell SA, et al. Plasmodium falciparum SERA5 plays a non-enzymatic role in the malarial asexual blood-stage lifecycle. Mol Microbiol. 2015; 96(2):368–87. https://doi.org/10.1111/mmi.12941 PMID: 25599609.
- Thomas JA, Collins CR, Das S, Hackett F, Graindorge A, Bell D, et al. Development and Application of a Simple Plaque Assay for the Human Malaria Parasite Plasmodium falciparum. PloS one. 2016; 11(6): e0157873. https://doi.org/10.1371/journal.pone.0157873 PMID: 27332706.
- Glushakova S, Mazar J, Hohmann-Marriott MF, Hama E, Zimmerberg J. Irreversible effect of cysteine protease inhibitors on the release of malaria parasites from infected erythrocytes. Cell Microbiol. 2009; 11(1):95–105. https://doi.org/10.1111/j.1462-5822.2008.01242.x PMID: 19016793; PubMed Central PMCID: PMC2883916.
- Li J, Mitamura T, Fox BA, Bzik DJ, Horii T. Differential localization of processed fragments of Plasmodium falciparum serine repeat antigen and further processing of its N-terminal 47 kDa fragment. Parasitol Int. 2002; 51(4):343–52. PMID: 12421632.
- Ruecker A, Shea M, Hackett F, Suarez C, Hirst EM, Milutinovic K, et al. Proteolytic activation of the essential parasitophorous vacuole cysteine protease SERA6 accompanies malaria parasite egress from its host erythrocyte. J Biol Chem. 2012; 287(45):37949–63. https://doi.org/10.1074/jbc.M112. 400820 PMID: 22984267; PubMed Central PMCID: PMC3488066.
- Schmidt-Christensen A, Sturm A, Horstmann S, Heussler VT. Expression and processing of Plasmodium berghei SERA3 during liver stages. Cell Microbiol. 2008; 10(8):1723–34. https://doi.org/10.1111/j. 1462-5822.2008.01162.x PMID: 18419771; PubMed Central PMCID: PMCPMC2613260.
- Aly AS, Matuschewski K. A malarial cysteine protease is necessary for Plasmodium sporozoite egress from oocysts. J Exp Med. 2005; 202(2):225–30. https://doi.org/10.1084/jem.20050545 PMID: 16027235; PubMed Central PMCID: PMCPMC2213010.
- Debrabant A, Maes P, Delplace P, Dubremetz JF, Tartar A, Camus D. Intramolecular mapping of Plasmodium falciparum P126 proteolytic fragments by N-terminal amino acid sequencing. Mol Biochem Parasitol. 1992; 53(1–2):89–95. PMID: 1501648.
- Delplace P, Bhatia A, Cagnard M, Camus D, Colombet G, Debrabant A, et al. Protein p126: a parasitophorous vacuole antigen associated with the release of Plasmodium falciparum merozoites. Biol Cell. 1988; 64(2):215–21. PMID: 3067800.

- Delplace P, Dubremetz JF, Fortier B, Vernes A. A 50 kilodalton exoantigen specific to the merozoite release-reinvasion stage of Plasmodium falciparum. Mol Biochem Parasitol. 1985; 17(2):239–51. PMID: 3906394.
- Delplace P, Fortier B, Tronchin G, Dubremetz JF, Vernes A. Localization, biosynthesis, processing and isolation of a major 126 kDa antigen of the parasitophorous vacuole of Plasmodium falciparum. Mol Biochem Parasitol. 1987; 23(3):193–201. PMID: 3299083.
- Yeoh S, O'Donnell RA, Koussis K, Dluzewski AR, Ansell KH, Osborne SA, et al. Subcellular discharge of a serine protease mediates release of invasive malaria parasites from host erythrocytes. Cell. 2007; 131(6):1072–83. PMID: 18083098. https://doi.org/10.1016/j.cell.2007.10.049
- Collins CR, Hackett F, Strath M, Penzo M, Withers-Martinez C, Baker DA, et al. Malaria parasite cGMPdependent protein kinase regulates blood stage merozoite secretory organelle discharge and egress. PLoS Pathog. 2013; 9(5):e1003344. Epub 2013/05/16. https://doi.org/10.1371/journal.ppat.1003344 PMID: 23675297; PubMed Central PMCID: PMC3649973.
- Withers-Martinez C, Strath M, Hackett F, Haire LF, Howell SA, Walker PA, et al. The malaria parasite egress protease SUB1 is a calcium-dependent redox switch subtilisin. Nat Commun. 2014; 5:3726. https://doi.org/10.1038/ncomms4726 PMID: 24785947; PubMed Central PMCID: PMC4024747.
- Collins CR, Das S, Wong EH, Andenmatten N, Stallmach R, Hackett F, et al. Robust inducible Cre recombinase activity in the human malaria parasite Plasmodium falciparum enables efficient gene deletion within a single asexual erythrocytic growth cycle. Mol Microbiol. 2013; 88(4):687–701. https://doi. org/10.1111/mmi.12206 PMID: 23489321; PubMed Central PMCID: PMC3708112.
- Andenmatten N, Egarter S, Jackson AJ, Jullien N, Herman JP, Meissner M. Conditional genome engineering in Toxoplasma gondii uncovers alternative invasion mechanisms. Nat Methods. 2013; 10 (2):125–7. https://doi.org/10.1038/nmeth.2301 PMID: 23263690; PubMed Central PMCID: PMCPMC3605914.
- Jullien N, Goddard I, Selmi-Ruby S, Fina JL, Cremer H, Herman JP. Conditional transgenesis using Dimerizable Cre (DiCre). PloS one. 2007; 2(12):e1355. https://doi.org/10.1371/journal.pone.0001355 PMID: 18159238; PubMed Central PMCID: PMCPMC2131782.
- Jullien N, Sampieri F, Enjalbert A, Herman JP. Regulation of Cre recombinase by ligand-induced complementation of inactive fragments. Nucleic acids research. 2003; 31(21):e131. PMID: 14576331; PubMed Central PMCID: PMCPMC275488. https://doi.org/10.1093/nar/gng131
- Ribacke U, Moll K, Albrecht L, Ahmed Ismail H, Normark J, Flaberg E, et al. Improved in vitro culture of Plasmodium falciparum permits establishment of clinical isolates with preserved multiplication, invasion and rosetting phenotypes. PloS one. 2013; 8(7):e69781. https://doi.org/10.1371/journal.pone.0069781 PMID: 23894537; PubMed Central PMCID: PMCPMC3718792.
- Taylor HM, McRobert L, Grainger M, Sicard A, Dluzewski AR, Hopp CS, et al. The malaria parasite cyclic GMP-dependent protein kinase plays a central role in blood-stage schizogony. Eukaryot Cell. 2010; 9(1):37–45. Epub 2009/11/17. <u>https://doi.org/10.1128/EC.00186-09</u> PMID: <u>19915077</u>; PubMed Central PMCID: PMC2805293.
- Hale VL, Watermeyer JM, Hackett F, Vizcay-Barrena G, van Ooij C, Thomas JA, et al. Parasitophorous vacuole poration precedes its rupture and rapid host erythrocyte cytoskeleton collapse in Plasmodium falciparum egress. Proc Natl Acad Sci U S A. 2017. https://doi.org/10.1073/pnas.1619441114 PMID: 28292906.
- Das S, Hertrich N, Perrin AJ, Withers-Martinez C, Collins CR, Jones ML, et al. Processing of Plasmodium falciparum Merozoite Surface Protein MSP1 Activates a Spectrin-Binding Function Enabling Parasite Egress from RBCs. Cell Host Microbe. 2015; 18(4):433–44. https://doi.org/10.1016/j.chom.2015. 09.007 PMID: 26468747; PubMed Central PMCID: PMCPMC4608996.
- Glushakova S, Yin D, Li T, Zimmerberg J. Membrane transformation during malaria parasite release from human red blood cells. Current biology: CB. 2005; 15(18):1645–50. <u>https://doi.org/10.1016/j.cub.</u> 2005.07.067 PMID: 16169486.
- **29.** Gilson PR, Crabb BS. Morphology and kinetics of the three distinct phases of red blood cell invasion by Plasmodium falciparum merozoites. Int J Parasitol. 2009; 39(1):91–6. <u>https://doi.org/10.1016/j.ijpara.</u> 2008.09.007 PMID: 18952091.
- Abkarian M, Massiera G, Berry L, Roques M, Braun-Breton C. A novel mechanism for egress of malarial parasites from red blood cells. Blood. 2011; 117(15):4118–24. <u>https://doi.org/10.1182/blood-2010-08-</u> 299883 PMID: 21297002.
- Dvorak JA, Miller LH, Whitehouse WC, Shiroishi T. Invasion of erythrocytes by malaria merozoites. Science. 1975; 187(4178):748–50. PMID: 803712.
- Glushakova S, Humphrey G, Leikina E, Balaban A, Miller J, Zimmerberg J. New stages in the program of malaria parasite egress imaged in normal and sickle erythrocytes. Current biology: CB. 2010; 20

(12):1117–21. https://doi.org/10.1016/j.cub.2010.04.051 PMID: 20537541; PubMed Central PMCID: PMC3541015.

- 33. Garg S, Agarwal S, Kumar S, Yazdani SS, Chitnis CE, Singh S. Calcium-dependent permeabilization of erythrocytes by a perforin-like protein during egress of malaria parasites. Nat Commun. 2013; 4:1736. Epub 2013/04/18. https://doi.org/10.1038/ncomms2725 PMID: 23591903.
- Aoki S, Li J, Itagaki S, Okech BA, Egwang TG, Matsuoka H, et al. Serine repeat antigen (SERA5) is predominantly expressed among the SERA multigene family of Plasmodium falciparum, and the acquired antibody titers correlate with serum inhibition of the parasite growth. J Biol Chem. 2002; 277 (49):47533–40. https://doi.org/10.1074/jbc.M207145200 PMID: 12244052.
- Bozdech Z, Llinas M, Pulliam BL, Wong ED, Zhu J, DeRisi JL. The transcriptome of the intraerythrocytic developmental cycle of Plasmodium falciparum. PLoS Biol. 2003; 1(1):E5. <u>https://doi.org/10.1371/journal.pbio.0000005</u> PMID: <u>12929205</u>; PubMed Central PMCID: PMCPMC176545.
- O'Donnell RA, Preiser PR, Williamson DH, Moore PW, Cowman AF, Crabb BS. An alteration in concatameric structure is associated with efficient segregation of plasmids in transfected Plasmodium falciparum parasites. Nucleic acids research. 2001; 29(3):716–24. PMID: <u>11160894</u>; PubMed Central PMCID: PMCPMC30406.
- O'Donnell RA, Freitas-Junior LH, Preiser PR, Williamson DH, Duraisingh M, McElwain TF, et al. A genetic screen for improved plasmid segregation reveals a role for Rep20 in the interaction of Plasmodium falciparum chromosomes. EMBO J. 2002; 21(5):1231–9. https://doi.org/10.1093/emboj/21.5.1231 PMID: 11867551; PubMed Central PMCID: PMCPMC125903.
- Callan-Jones A, Albarran Arriagada OE, Massiera G, Lorman V, Abkarian M. Red blood cell membrane dynamics during malaria parasite egress. Biophys J. 2012; 103(12):2475–83. https://doi.org/10.1016/j. bpj.2012.11.008 PMID: 23260049; PubMed Central PMCID: PMCPMC3525858.
- Lew VL. Malaria: surprising mechanism of merozoite egress revealed. Current biology: CB. 2011; 21 (9):R314–6. https://doi.org/10.1016/j.cub.2011.03.066 PMID: 21549952.
- 40. Chandramohanadas R, Davis PH, Beiting DP, Harbut MB, Darling C, Velmourougane G, et al. Apicomplexan parasites co-opt host calpains to facilitate their escape from infected cells. Science. 2009; 324 (5928):794–7. Epub 2009/04/04. <u>https://doi.org/10.1126/science.1171085</u> PMID: <u>19342550</u>; PubMed Central PMCID: PMC3391539.
- Yang AS, O'Neill MT, Jennison C, Lopaticki S, Allison CC, Armistead JS, et al. Cell traversal activity is important for Plasmodium falciparum liver infection in humanized mice. Cell Rep. 2017; 18(13):3105– 3116. https://doi.org/10.1016/j.celrep.2017.03.017 PMID: 28355563.
- Yahata K, Treeck M, Culleton R, Gilberger TW, Kaneko O. Time-lapse imaging of red blood cell invasion by the rodent malaria parasite Plasmodium yoelii. PloS one. 2012; 7(12):e50780. https://doi.org/ 10.1371/journal.pone.0050780 PMID: 23227208; PubMed Central PMCID: PMCPMC3515438.
- Riglar DT, Richard D, Wilson DW, Boyle MJ, Dekiwadia C, Turnbull L, et al. Super-resolution dissection of coordinated events during malaria parasite invasion of the human erythrocyte. Cell Host Microbe. 2011; 9(1):9–20. Epub 2011/01/18. https://doi.org/10.1016/j.chom.2010.12.003 PMID: 21238943.
- Crick AJ, Theron M, Tiffert T, Lew VL, Cicuta P, Rayner JC. Quantitation of malaria parasite-erythrocyte cell-cell interactions using optical tweezers. Biophys J. 2014; 107(4):846–53. Epub 2014/08/21. https:// doi.org/10.1016/j.bpj.2014.07.010 PMID: 25140419; PubMed Central PMCID: PMC4142257.
- Boyle MJ, Wilson DW, Richards JS, Riglar DT, Tetteh KK, Conway DJ, et al. Isolation of viable Plasmodium falciparum merozoites to define erythrocyte invasion events and advance vaccine and drug development. Proc Natl Acad Sci U S A. 2010; 107(32):14378–83. https://doi.org/10.1073/pnas.1009198107 PMID: 20660744; PubMed Central PMCID: PMCPMC2922570.
- 46. Adrain C, Freeman M. New lives for old: evolution of pseudoenzyme function illustrated by iRhoms. Nat Rev Mol Cell Biol. 2012; 13(8):489–98. Epub 2012/07/12. <u>https://doi.org/10.1038/nrm3392</u> PMID: 22781900.
- Pils B, Schultz J. Inactive enzyme-homologues find new function in regulatory processes. J Mol Biol. 2004; 340(3):399–404. Epub 2004/06/24. https://doi.org/10.1016/j.jmb.2004.04.063 PMID: 15210342.
- **48.** Todd AE, Orengo CA, Thornton JM. Sequence and structural differences between enzyme and nonenzyme homologs. Structure. 2002; 10(10):1435–51. PMID: 12377129.
- Zeqiraj E, van Aalten DM. Pseudokinases-remnants of evolution or key allosteric regulators? Curr Opin Struct Biol. 2010; 20(6):772–81. https://doi.org/10.1016/j.sbi.2010.10.001 PMID: 21074407; PubMed Central PMCID: PMCPMC3014569.
- 50. Willert EK, Fitzpatrick R, Phillips MA. Allosteric regulation of an essential trypanosome polyamine biosynthetic enzyme by a catalytically dead homolog. Proc Natl Acad Sci U S A. 2007; 104(20):8275–80. https://doi.org/10.1073/pnas.0701111104 PMID: 17485680; PubMed Central PMCID: PMCPMC1895940.

- Etheridge RD, Alaganan A, Tang K, Lou HJ, Turk BE, Sibley LD. The Toxoplasma pseudokinase ROP5 forms complexes with ROP18 and ROP17 kinases that synergize to control acute virulence in mice. Cell Host Microbe. 2014; 15(5):537–50. https://doi.org/10.1016/j.chom.2014.04.002 PMID: 24832449; PubMed Central PMCID: PMCPMC4086214.
- Holder AA, Freeman RR. Biosynthesis and processing of a Plasmodium falciparum schizont antigen recognized by immune serum and a monoclonal antibody. J Exp Med. 1982; 156(5):1528–38. PMID: 6752328; PubMed Central PMCID: PMC2186849.
- **53.** Blackman MJ, Whittle H, Holder AA. Processing of the Plasmodium falciparum major merozoite surface protein-1: identification of a 33-kilodalton secondary processing product which is shed prior to erythrocyte invasion. Mol Biochem Parasitol. 1991; 49(1):35–44. PMID: 1723148.
- 54. Withers-Martinez C, Saldanha JW, Ely B, Hackett F, O'Connor T, Blackman MJ. Expression of recombinant Plasmodium falciparum subtilisin-like protease-1 in insect cells: Characterization, comparison with the parasite protease, and homology modelling. J Biol Chem. 2002; 277:29698–709. PMID: <u>12052828</u>. <u>https://doi.org/10.1074/jbc.M203088200</u>
- Blackman MJ. Purification of Plasmodium falciparum merozoites for analysis of the processing of merozoite surface protein-1. Methods Cell Biol. 1994; 45:213–20. PMID: 7707987.
- Harris PK, Yeoh S, Dluzewski AR, O'Donnell RA, Withers-Martinez C, Hackett F, et al. Molecular identification of a malaria merozoite surface sheddase. PLoS Pathog. 2005; 1(3):241–51. PMID: 16322767. https://doi.org/10.1371/journal.ppat.0010029
- Collins CR, Withers-Martinez C, Hackett F, Blackman MJ. An inhibitory antibody blocks interactions between components of the malarial invasion machinery. PLoS Pathog. 2009; 5(1):e1000273. https:// doi.org/10.1371/journal.ppat.1000273 PMID: 19165323; PubMed Central PMCID: PMC2621342.
- Moss DK, Remarque EJ, Faber BW, Cavanagh DR, Arnot DE, Thomas AW, et al. Plasmodium falciparum 19-kilodalton merozoite surface protein 1 (MSP1)-specific antibodies that interfere with parasite growth in vitro can inhibit MSP1 processing, merozoite invasion, and intracellular parasite development. Infect Immun. 2012; 80(3):1280–7. Epub 2011/12/29. https://doi.org/10.1128/IAI.05887-11 PMID: 22202121; PubMed Central PMCID: PMC3294643.
- Blackman MJ, Fujioka H, Stafford WH, Sajid M, Clough B, Fleck SL, et al. A subtilisin-like protein in secretory organelles of Plasmodium falciparum merozoites. J Biol Chem. 1998; 273(36):23398–409. PMID: 9722575.